

DEPARTMENT OF ECONOMICS AND BUSINESS ECONOMICS AARHUS UNIVERSITY



Bond Market Asymmetries across Recessions and Expansions: New Evidence on Risk Premia

Martin M. Andreasen, Tom Engsted, Stig V. Møller and Magnus Sander

CREATES Research Paper 2016-26

Department of Economics and Business Economics Aarhus University Fuglesangs Allé 4 DK-8210 Aarhus V Denmark Email: oekonomi@au.dk Tel: +45 8716 5515

Bond Market Asymmetries across Recessions and Expansions: New Evidence on Risk Premia^{*}

Martin M. Andreasen[†] Tom Engsted[‡] Stig V. Møller[§] Magnus Sander[¶] Aarhus University and CREATES

August 30, 2016

Abstract

This paper provides new evidence on bond risk premia by conditioning the classic Campbell-Shiller regressions on the business cycle. In expansions, we find mostly positive intercepts and negative regression slopes, but the results are completely reversed in recessions with *negative* intercepts and *positive* regression slopes. We reproduce these coefficients in a term structure model with business cycle dependent loadings in the market price of risk. This model also predicts excess returns in the right direction during expansions and recessions, whereas the Gaussian affine term structure model predicts excess returns for medium- and long-term bonds with the wrong sign during recessions.

Keywords: Bond return predictability, Business cycle variation in excess returns, Market price of risk, Zero-lower bound, Unspanned macroeconomic risk.

JEL: E43, E44, G12.

^{*}We thank Jens Christensen, Leonardo Iania, and Robin Lumsdaine for useful comments and discussions. Remarks and suggestions from seminar participants at Louvain School of Management (CORE) are also much appreciated. We acknowledge access to computer facilities provided by the Danish Center for Scientific Computing (DCSC). We also acknowledge support from CREATES - Center for Research in Econometric Analysis of Time Series (DNRF78), funded by the Danish National Research Foundation.

[†]Fuglesangs Allé 4, 8210 Aarhus V, Denmark, email: mandreasen@econ.au.dk, telephone +45 87165982.

[‡]Fuglesangs Allé 4, 8210 Aarhus V, Denmark, email: tengsted@econ.au.dk, telephone +45 87165572.

[§]Fuglesangs Allé 4, 8210 Aarhus V, Denmark, email: svm@econ.au.dk, telephone +45 87164825.

[¶]Fuglesangs Allé 4, 8210 Aarhus V, Denmark, email: mdsj@econ.au.dk, telephone +45 87166021.

1 Introduction

The expectations hypothesis of the term structure specifies that any long-term interest rate is given by the average of expected future short rates and that excess returns on any long-term bond therefore are unpredictable. Despite its appealing intuition, the expectations hypothesis is often rejected empirically, perhaps most forcefully in the seminal study of Campbell and Shiller (1991). This paper provides new evidence on bond risk premia by conditioning the classic Campbell-Shiller regressions on the business cycle through separate intercepts and slope coefficients in expansions and recessions. This extension is motivated by previous research showing that interest rates are more persistent in expansions than in recessions and that two-state models describe interest rate dynamics much better than single-state models, see Hamilton (1988), Gray (1996), Ang and Bekaert (2002), Bansal and Zhou (2002), among others.

Our modified Campbell-Shiller regressions reveal a unique business cycle dependent pattern in the relation between the slope of the yield curve (i.e., the yield spread) and subsequent yield changes. In expansions, we find mostly positive intercepts and negative regression slopes that decrease with maturity, similar to what typically is reported for ordinary Campbell-Shiller regressions, i.e. without conditioning on the business cycle (see, e.g., Bekaert, Hodrick and Marshall (1997) and Dai and Singleton (2002)). By contrast, in recessions, we obtain *negative* intercepts and *positive* slope coefficients that generally increase with maturity. This switch in the regression coefficients is significant using both asymptotic and bootstrapped Wald tests. Given that the slope coefficients are closer to the expectations hypothesis in recessions than in expansions, this novel result suggests that bond risk premia, as measured by excess returns, are more predictable by the yield spread during expansions than during recessions. We also show that the evidence of asymmetric return predictability extends to other prominent yield-based predictors such as the forward spread (Fama and Bliss (1987)) and the return forecasting factor introduced by Cochrane and Piazzesi (2005).¹

To provide an explanation of these new empirical findings, we propose a dynamic term structure model (DTSM) with business cycle dependent loadings in the market price of risk. Contrary to previous DTSMs with regime switching, we discipline our model to have observed regimes of either expansions or recessions but include time-varying physical transition probabilities between

¹Hence, forecasting bond returns during recessions from macro variables as in Ludvigson and Ng (2009) seems more promising, although Duffee (2013) and Bauer and Hamilton (2015) question the ability of macro variables to predict bond returns beyond the information contained in the yield curve when accounting for small-sample distortions.

regimes, which may depend on the business cycle and the current yield curve. To ensure fast and reliable inference of our model, we extend the sequential regression (SR) approach of Andreasen and Christensen (2015) to estimate all parameters in our regime-dependent market price of risk in closed form by a modified OLS regression. This implies that our regime-switching model can be estimated in a few minutes with no additional computational costs compared to the Gaussian ATSM.

The main results from estimating our regime-switching model with three pricing factors on monthly U.S. data (1961:6 to 2013:12) are as follows. First, our model is able to match the empirical intercepts and slope coefficients in ordinary Campbell-Shiller regressions and in our modified Campbell-Shiller regressions conditioning on the business cycle. As a consequence, model-implied expected excess returns correlate positively with realized excess returns during both expansions and recessions. In contrast, the inability of the Gaussian ATSM to match the switch in the Campbell-Shiller regression slopes implies that this model predicts excess returns for medium- and long-term bonds with the wrong sign during recessions. Second, our model also replicates the observed asymmetry in return predictability, i.e. that the yield spread, the forward spread, and the Cochrane-Piazzesi factor have strong predictive power in expansions but not in recessions. Third, to account for these asymmetric patterns in the yield curve across the business cycle, our model generates a *negative* relation between the short rate and excess returns in expansions (as in the Gaussian ATSM), but a *positive* relation in recessions. This means that our model displays a tendency for expected excess returns to peak at the start of a recession and then mean-revert during the middle and last part of a recession when the Federal Reserve starts to lower its policy rate. Thus, our model suggests that the Federal Reserve is able to remove some of the risks attached to recessions, as accommodating monetary policy reduces the required risk compensation in the bond market. This effect of monetary policy is not present in the Gaussian ATSM, which generally predicts that expected excess returns increase throughout the entire recession. We finally show that our fully flexible regimeswitching model is robust to accounting for the zero lower bound and may be simplified to only have two parameters in the market price of risk that switch between expansions and recessions.

The rest of the paper is organized as follows. Section 2 presents new empirical evidence on bond risk premia from modified Campbell-Shiller and return regressions. We then introduce a new DTSM with regime-switching in Section 3 and explain how to estimate this model by the SR approach. Section 4 explores the ability of this model to replicate the empirical findings from Section 2, while Section 5 examines the robustness of our model and relates it to the existing literature. Section 6 concludes, while Appendix A contains some econometric details related to our analysis.²

2 New Empirical Evidence on Bond Risk Premia

This section presents new evidence on the business cycle properties of bond risk premia. We present our main empirical finding in Sections 2.1 and 2.2 and explore its robustness in Section 2.3. The main implication of our new empirical results for DTSMs are discussed in Section 2.4.

2.1 Modified Campbell-Shiller Regressions

To motivate our new empirical finding, consider the ordinary Campbell and Shiller (1991) regression

$$y_{t+m,k-m} - y_{t,k} = \alpha_k + \beta_k \frac{m}{k-m} \left(y_{t,k} - y_{t,m} \right) + u_{t+m,k},\tag{1}$$

where $y_{t,k}$ refers to the k-period bond yield in period t. As is common in the literature, we set one period equal to one month and implement (1) by running the regressions with m = 3 for k = 6, 9, 12, ..., 120, implying a total of 39 regressions. That is, $y_{t,m}$ in (1) corresponds to the three month interest rate. The upper part of Figure 1 shows the results of estimating (1) from 1961:6 to 2013:12, with intercepts on the left and slope estimates on the right. The full and dashed lines are based on Fama and Bliss (1987) and Gürkaynak, Sack and Wright (2007) bond yields, respectively.³ According to the expectations hypothesis $y_{t,k}^{EH} = \frac{1}{k} \sum_{i=0}^{k-1} \mathbb{E}_t [y_{t+i,1}] + c_k$, we should see $\beta_k = 1$ for all k. However, as in Campbell and Shiller (1991), the slope estimates are negative and decreasing with maturity, constituting a clear violation of the expectations hypothesis.

Now consider what happens when conditioning on the state of business cycle. That is, we run the Campbell-Shiller regressions interacted with business cycle dummies, i.e.

$$y_{t+m,k-m} - y_{t,k} = \alpha_k^{EXP} \mathbf{1}_{\{z_t \ge c\}} + \beta_k^{EXP} \frac{m}{k-m} \mathbf{1}_{\{z_t \ge c\}} (y_{t,k} - y_{t,m})$$

$$+ \alpha_k^{REC} \left(1 - \mathbf{1}_{\{z_t \ge c\}} \right) + \beta_k^{REC} \frac{m}{k-m} \left(1 - \mathbf{1}_{\{z_t \ge c\}} \right) (y_{t,k} - y_{t,m}) + \tilde{u}_{t+m,k},$$

$$(2)$$

²In addition, detailed data descriptions, robustness checks, and all model derivations are provided in an Online Appendix, which is available from the authors' homepages or upon request.

³See Appendix A.1 for further details on the data.

where $1_{\{z_t \ge c\}}$ is the indicator function with a value of one for expansions when $z_t \ge c$ and zero otherwise. We measure recessions by letting z_t refer to the Purchasing Managers' Index (PMI) and use the threshold value c = 44.5 from Berge and Jordà (2011) to identify recessions and expansions. The PMI is a widely watched indicator of business cycle activity and has the advantage of being available in real time without publication lags or subsequent data revisions. Furthermore, Christiansen, Eriksen and Møller (2014) provide empirical evidence that the PMI is the single best recession indicator among a large panel of economic variables. As shown below, our results are robust to using standard NBER recession dates, although they are subject to publication lags and therefore not our preferred recession indicator.

The lower part of Figure 1 summarizes the results of estimating (2). In expansions, the estimated slopes β_k^{EXP} and intercepts α_k^{EXP} are broadly similar to those obtained in the ordinary Campbell-Shiller regressions. In recessions, however, the slope coefficients β_k^{REC} are *positive* and mostly increasing with maturity. At long maturities they are even above one. In addition, the intercepts α_k^{REC} are negative at all maturities which under the expectations hypothesis is consistent with an upward sloping yield curve.⁴ Hence, in recessions the slope of the yield curve predicts future long-term bond yields in the direction implied by the expectations hypothesis, whereas the opposite holds during expansions.

Although recessions typically are short lived, allowing for the sign switch has a large effect on the goodness-of-fit in the Campbell-Shiller regressions. This is illustrated in Figure 2, which shows that the R^2 statistic increases from around 3% in the ordinary Campbell-Shiller regressions to between 5% and 6% when conditioning on the business cycle.

Having outlined economically interesting differences in the Campbell-Shiller loadings across recessions and expansions, a natural next step is to test whether these differences are statistically significant. Table 1 shows Wald statistics that test for joint equality between expansion and recession coefficients across maturities. We report both asymptotic and bootstrap p-values. Appendices A.2 to A.4 describe the construction of the Wald test and the bootstrap procedure. To also explore robustness, we report results for the full sample and two subsamples: 1973:1-2013:12 (the panel of bond yields is balanced starting from 1973:1) and 1983:1-2013:12 (the post-Volcker period). To

⁴The derivation is provided in the Online Appendix. The pure version of the expectations hypothesis, i.e. $y_{t,k}^{pure \ EH} = \frac{1}{k} \sum_{i=0}^{k-1} \mathbb{E}_t [y_{t+i,1}]$, also restricts the intercept to zero, so that expected bond returns across all maturities are equal, i.e. $\alpha_k = 0$ for all k. However, the literature typically only focuses on whether bond risk premia are constant over time, i.e. whether $\beta_k = 1$ for all k.

identify recessions and expansions, we consider both the PMI and the NBER recession dates. The results of the Wald tests are clear-cut. The null hypothesis that expansion and recession slopes coefficients are equal, i.e. $\beta_k^{REC} = \beta_k^{EXP}$ for all k, is strongly rejected at any conventional significance level. Similarly, a Wald test also rejects the null that $\alpha_k^{REC} = \alpha_k^{EXP}$ for all k. Hence, the observed change in the intercepts and the slope coefficients across expansions and recessions are statistically significant. The Online Appendix also provides Wald tests of the expectations hypothesis, which is rejected in both regimes.⁵

2.2Modified Return Regressions

Our modified Campbell-Shiller regressions in (2) show that the slope coefficients during recessions generally are closer to the expectations hypothesis than the corresponding loadings in expansions. As excess returns are unpredictable under the expectations hypothesis, we therefore find higher return predictability by the yield spread during expansions than in recessions.⁶ But does this pattern also hold for other classic yield-based predictors of bond returns such as the forward spread (Fama (1976); Fama and Bliss (1987)) and the Cochrane and Piazzesi (2005) (CP) return forecasting factor?

We address this question by running standard univariate return regressions for three-months ahead excess returns

$$xhpr_{t+m,k} = \mu_k + \theta_k x_{t,k} + e_{t+m,k},\tag{3}$$

and its modified version conditioning on the business cycle, i.e.

$$xhpr_{t+m,k} = \mu_k^{EXP} \mathbf{1}_{\{z_t \ge c\}} + \theta_k^{EXP} \mathbf{1}_{\{z_t \ge c\}} x_{t,k}$$

$$+ \mu_k^{REC} \left(1 - \mathbf{1}_{\{z_t \ge c\}} \right) + \theta_k^{REC} \left(1 - \mathbf{1}_{\{z_t \ge c\}} \right) x_{t,k} + \tilde{e}_{t+m,k}.$$

$$(4)$$

Here, $xhpr_{t+m,k} \equiv hpr_{t+m,k} - \frac{m}{12}y_{t,m}$ and the holding period return $hpr_{t+m,n} \equiv -\frac{k-m}{12}y_{t+m,k-m} + \frac{m}{12}y_{t+m,k-m}$ $\frac{k}{12}y_{t,k}$, whereas $x_{t,k}$ may either refer to the yield spread $y_{t,k} - y_{t,m}$, the forward spread, or the CP $factor.^7$

⁵Althought the slope coefficients are positive and closer to one in recessions, their values are substantially larger than one at long maturities and this explains why the expectations hypothesis also is rejected in recessions.

⁶Here, we exploit the one-to-one relation between Campbell-Shiller regressions and regressing excess returns on

the yield spread. The exact expression for this relation is provided in our Online Appendix. ⁷The forward spread is $f_t^{(k-m,k)} - \frac{m}{12}y_{t,m}$, where $f_t^{(k-m,k)} \equiv \frac{k}{12}y_{t,k} - \frac{k-m}{12}y_{t,k-m}$ is the forward rate between time t + k - m and t + k. The CP factor is given by $cp_t = \hat{\gamma}' F_t$, where $F_t \equiv \begin{bmatrix} 1, y_{t,12}, f_t^{(12,24)}, f_t^{(36,48)}, f_t^{(36,48)}, f_t^{(48,60)} \end{bmatrix}'$

The upper panel of Figure 3 uses the yield spread in (3) and (4) and shows the expected pattern in bond return predictability across the business cycle. That is, θ_k^{EXP} is positive in expansions and generally increases with maturity, whereas θ_k^{REC} is negative in recessions and generally decreases with maturity. The middle and lower part of Figure 3 show that the forward spread and the CP factor, respectively, also imply different degrees of predictability for bond returns across the business cycle, with regression slopes mostly having the 'wrong' sign in recessions.

To summarize, the return regressions reveal a striking pattern in the slope coefficients across the business cycle. In expansions we find the typical pattern of positive slopes on the yield spread, the forward spread, and the CP factor, while in recessions the slopes are either close to zero or even negative.

2.3 Robustness

2.3.1 Extending the Sample and the Maturity Range

Since our empirical findings are new, we have done a wide range of robustness checks. First of all, we extend the sample back to 1926:1 in the top part of Table 2 to increase the number of observations, where the economy is in recession. The considered return series is for a 10-year government bond provided by Global Financial Data. Regimes are here identified from the NBER recessions dates, as the PMI indicator is unavailable over this longer sample. For the period 1926:1-2013:12, $y_{t,k} - y_{t,m}$ is statistically significant and explains 5.3% of the variation in excess bond returns. The R^2 statistic increases to 7.7% by allowing the regression coefficients to change across expansions and recessions. In expansions, the slope coefficient is positive and strongly statistically significant. Results from different subsamples confirm a significant positive relation between the slope of the yield curve and subsequent excess bond returns in expansions, while in recessions the relation is insignificant.

Second, we next extend the maturity range to 20 years by using long-term government bond returns from Ibbotson and a related measure of the 20-year yield spread. The second part of Table 2 reveals the same pattern as above, with strong predictability in expansions but not in recessions. Hence, the systematic difference in bond predictability across the business cycle is not limited to the 10-year maturity range, although this is the main focus of our paper.

and $\hat{\gamma}$ is obtained by regressing one-year ahead excess bond returns on F_t , i.e. $\frac{1}{4} \sum_{i=2}^5 rx_{t+12,i12} = \gamma' F_t + \overline{e}_{t+12}$.

2.3.2 Speed of Transition, Out-of-Sample Evidence, and Small-Sample Biases

We have also investigated whether the switch in (2) and (4) between recessions and expansions is best modeled using a smoothly changing function or the binary specification based on dummy variables used above. In smooth transition regressions with the logistic function, we find that the transition is practically instant. In addition, out-of-sample regressions show that all three yieldbased predictors (i.e., the yield spread, the forward spread, and the CP factor) only imply significant bond return predictability in expansions. Finally, we have checked that our results are robust to small-sample biases using a stationary bootstrap. The details of all these robustness checks are delegated to our Online Appendix.

2.4 Implication for DTSMs

There is a strong conclusion to draw from the above empirical evidence: Bond market asymmetries across the business cycle are substantial and economically important. In particular, the predictive power of the yield spread for future bond yields and bond returns crucially depends on the state of business cycle. To reproduce the negative regression loadings from the ordinary Campbell-Shiller regressions in (1), most DTSMs require a negative correlation between the short rate and excess returns according to Dai and Singleton (2002). Thus, to explain the sign switch in regression loadings from our modified Campbell-Shiller regressions in (2), a successful model should generate a *negative* relation between the short rate and excess returns in expansions, but a *positive* relation in recessions.

3 A DTSM with a Regime-Dependent Market Price of Risk

We next explore whether a DTSM can reproduce the documented asymmetries in the U.S. yield curve across the business cycle, while at the same time matching properties of bond risk premia unrelated to the business cycle as given by (1) and (3). Dai and Singleton (2002) show that the Gaussian ATSM can reproduce loadings from ordinary Campbell-Shiller regressions, and this model therefore serves as a natural starting point. We proceed by motivating our extension of the Gaussian ATSM in Section 3.1, before formally presenting our proposed model in Section 3.2. Estimation of this model by the SR approach is briefly described in Section 3.3. We finally relate our proposed DTSM to the existing literature in Section 3.4.

3.1 Motivation

This section explores whether a further generalization of the market price of risk in the Gaussian ATSM has the potential to reproduce the documented asymmetries in the U.S. yield curve. To introduce our notation for this model, the short rate is given by

$$r_t = \alpha + \boldsymbol{\beta}' \mathbf{x}_t,\tag{5}$$

where α is a scalar and β is an $n_x \times 1$ vector. The dynamics of the n_x pricing factors under the risk-neutral measure \mathbb{Q} is specified as

$$\mathbf{x}_{t+1} = (\mathbf{I} - \mathbf{\Phi}) \, \mathbf{x}_t + \boldsymbol{\Sigma}_{\mathbf{x}} \boldsymbol{\varepsilon}_{\mathbf{x}, t+1}^{\mathbb{Q}}$$
(6)

with $\boldsymbol{\varepsilon}_{\mathbf{x},t+1}^{\mathbb{Q}} \sim \mathcal{NID}(\mathbf{0}, \mathbf{I})$.⁸ The \mathbb{P} dynamics follow from an essential affine market price of risk as in Duffee (2002), i.e. $\boldsymbol{\lambda}_t = \boldsymbol{\Sigma}_{\mathbf{x}}^{-1} (\boldsymbol{\lambda}_0 + \boldsymbol{\lambda}_{\mathbf{x}} \mathbf{x}_t)$. In the absence of arbitrage, the price of a zero-coupon bond in period t with maturity k is then given by $P_{t,k} = \exp\{A_k + \mathbf{B}'_k \mathbf{x}_t\}$, where the recursive expressions for A_k and \mathbf{B}_k are easily derived. Provided that the \mathbb{P} distribution for \mathbf{x}_t is stationary, the ordinary Campbell-Shiller coefficients in the Gaussian ATSM are then given by (with m = 1):⁹

$$\beta_{k} = 1 + \frac{\left(\frac{1}{k}\mathbf{B}_{k}' + \boldsymbol{\beta}'\right)\mathbb{V}\left[\mathbf{x}_{t}\right]\boldsymbol{\lambda}_{\mathbf{x}}'\mathbf{B}_{k-1}}{\left(\frac{1}{k}\mathbf{B}_{k}' + \boldsymbol{\beta}'\right)\mathbb{V}\left[\mathbf{x}_{t}\right]\left(\frac{1}{k}\mathbf{B}_{k} + \boldsymbol{\beta}\right)}$$
(7)

$$\alpha_{k} = -\frac{1}{k-1}A_{k-1} + \frac{1}{k}A_{k} - \hat{\beta}_{k}\mathbb{E}\left[\frac{y_{t,k} - y_{t,1}}{k-1}\right]$$

$$-\frac{1}{k-1}\left(\mathbf{B}_{k-1}'\left(\boldsymbol{\lambda}_{0} + \boldsymbol{\lambda}_{\mathbf{x}}\mathbb{E}\left[\mathbf{x}_{t}\right]\right) + \left(\frac{1}{k}\mathbf{B}_{k}' + \boldsymbol{\beta}'\right)\mathbb{E}\left[\mathbf{x}_{t}\right]\right).$$

$$(8)$$

Now suppose the average U.S. bond investor re-prices risk across the business cycle in such a way that $\lambda_{\mathbf{x}}$ has different values in expansions and recessions. According to (7), this modification has the potential to generate negative loadings of $\hat{\beta}_k$ in expansions and positive loadings in recessions. Equation (8) shows that a switch in $\lambda_{\mathbf{x}}$ between expansions and recessions also affects the intercept

⁸The intercept in the \mathbb{Q} distribution for \mathbf{x}_t is normalized to zero.

⁹The derivations are provided in our Online Appendix.

 $\hat{\alpha}_k$ even if λ_0 is constant. However, it may be necessary to also let λ_0 switch between recessions and expansions to match the observed difference in $\hat{\alpha}_k^{REC}$ and $\hat{\alpha}_k^{EXP}$.

Thus, considering a regime-dependent market price of risk based on economic activity has the potential to explain the observed asymmetries in our modified Campbell-Shiller regressions or, equivalently, the asymmetric behavior of bond risk premia across the business cycle.

3.2 Model Description

We next formally present a DTSM with a regime-dependent market price of risk based on economic activity. As in the Gaussian ATSM, the dynamics of the short rate under \mathbb{Q} is given by (5) and (6), implying that zero-coupon bond prices and the yield curve have the same expression as in the Gaussian ATSM, i.e.

$$y_{t,k} = \hat{A}_k + \hat{\mathbf{B}}'_k \mathbf{x}_t,\tag{9}$$

with $\tilde{A}_k \equiv \frac{-1}{k} A_k$ and $\tilde{\mathbf{B}}_k \equiv \frac{-1}{k} \mathbf{B}_k$ for k = 1, 2, ..., K.

We now deviate from the Gaussian ATSM by assuming that the market price of risk is piece-wise affine in the price factors \mathbf{x}_t , with loadings depending on whether the economy is in expansion or recession. That is, we let

$$\boldsymbol{\lambda}_{t} = \mathbf{1}_{\{z_{t} \geq c\}} \boldsymbol{\Sigma}_{\mathbf{x}}^{-1} \left(\boldsymbol{\lambda}_{0}^{(1)} + \boldsymbol{\lambda}_{\mathbf{x}}^{(1)} \mathbf{x}_{t} \right) + \left(1 - \mathbf{1}_{\{z_{t} \geq c\}} \right) \boldsymbol{\Sigma}_{\mathbf{x}}^{-1} \left(\boldsymbol{\lambda}_{0}^{(2)} + \boldsymbol{\lambda}_{\mathbf{x}}^{(2)} \mathbf{x}_{t} \right).$$
(10)

To discipline our model, the regimes are taken to be observed, whereas previous work mainly considers unobserved regimes. As in Section 2, z_t refers to the PMI and recessions are identified when z_t is below its threshold value of c = 44.5. Given that our model remains conditional Gaussian, a simple change of measure gives the following \mathbb{P} dynamics

$$\mathbf{x}_{t+1} = \mathbf{1}_{\{z_t \ge c\}} \boldsymbol{\lambda}_0^{(1)} + (1 - \mathbf{1}_{\{z_t \ge c\}}) \boldsymbol{\lambda}_0^{(2)} + \left(\mathbf{I} - \boldsymbol{\Phi} + \mathbf{1}_{\{z_t \ge c\}} \boldsymbol{\lambda}_{\mathbf{x}}^{(1)} + (1 - \mathbf{1}_{\{z_t \ge c\}}) \boldsymbol{\lambda}_{\mathbf{x}}^{(2)}\right) \mathbf{x}_t + \boldsymbol{\Sigma}_{\mathbf{x}} \boldsymbol{\epsilon}_{\mathbf{x}, t+1}^{\mathbb{P}}.$$
(11)

The model is closed by letting the \mathbb{P} dynamics of z_t evolve as

$$z_{t+1} = \gamma_0 + \gamma_z z_t + \boldsymbol{\gamma}'_{\mathbf{x}} \mathbf{x}_t + \Sigma_{zz} \epsilon^{\mathbb{P}}_{z,t+1}, \qquad (12)$$

where $\epsilon_{z,t+1}^{\mathbb{P}} \sim \mathcal{NTD}(0,1)$ and is independent of $\epsilon_{\mathbf{x},t+1}^{\mathbb{P}}$. That is, z_t may depend on its own lag if $\gamma_z \neq 0$ and lagged values of the pricing factors if $\gamma_{\mathbf{x}} \neq \mathbf{0}_{n_x \times 1}$. The latter implies that yield curve dynamics affect economic activity and hence introduce a feedback effect from financial markets to the real economy. If $\gamma_{\mathbf{x}} = \boldsymbol{\beta} \times \kappa$, where κ is some scalar, then (12) reduces to $z_{t+1} =$ $(\gamma_0 - \kappa \alpha) + \gamma_z z_t + \kappa r_t + \sum_{zz} \epsilon_{z,t+1}^{\mathbb{P}}$, implying that the short rate is a sufficient statistic for economic activity, as assumed in the standard New Keynesian model, see Woodford (2003). In the general case where $\gamma_{\mathbf{x}} \neq \mathbf{0}_{n_x \times 1}$, all yield curve dynamics as captured by the pricing factors \mathbf{x}_t may matter for economic activity.¹⁰ For instance, changes in long-term bond yields may have an independent effect on economic activity beyond variation in the short rate, as considered in some of the recent macroeconomic literature (see, for instance, Andrés, López-Salido and Nelson (2004) and Gerler and Karadi (2013)). Finally, if $\gamma_{\mathbf{x}} = \mathbf{0}_{n_x \times 1}$, then yield curve dynamics do not affect economic activity as assumed in Ang and Piazzesi (2003).

We also note from (11) that the switch in λ_t between recessions and expansions generates an instantaneous switch in the pricing factors and hence in the yield curve. This property of our model is thus consistent with our modified Campbell-Shiller and excess return regressions in Sections 2.1 and 2.2.

An interesting aspect of our model relates to the fact that z_t does not enter in the \mathbb{Q} distribution of the short rate and hence as a pricing factor in (9). This means that z_t is a 'hidden' factor as in Duffee (2011) and that our model displays unspanned macroeconomic risk similar to the work of Joslin, Priebsch and Singleton (2014). Hence, our macro variable z_t only affects yield curve dynamics indirectly by improving the forecast distribution of the pricing factors. Such improved predictions may be useful when forecasting the yield curve but also when computing expected excess returns or other measures of bond risk premia. Joslin, Priebsch and Singleton (2014) consider a setup where the macro variables enter linearly in the law of motion for the pricing factors under the \mathbb{P} measure, implying that their model may be characterized as displaying 'linear' unspanned macroeconomic risk. Our setup differs slightly from the one considered by Joslin, Priebsch and Singleton (2014), because the macro variable z_t in our model has a non-linear effect on the \mathbb{P} distribution for the pricing factors, meaning that our model may be described as having 'nonlinear' unspanned macroeconomic risk.

Finally, given that the \mathbb{Q} distribution of the short rate is identical to the one in the Gaussian

 $^{^{10}\}mathrm{A}$ similar specification is adopted in Diebold, Rudebusch and Aruoba (2006).

ATSM, our model implies the same identifying assumptions as the Gaussian ATSM with latent pricing factors. We require i) $\boldsymbol{\beta} = \mathbf{1}$, ii) $\boldsymbol{\Phi}$ to be diagonal with increasing eigenvalues, and iii) $\boldsymbol{\Sigma}_{\mathbf{x}}$ to be triangular.¹¹ This identification scheme constrains the \mathbb{Q} dynamics for the pricing factors whereas the \mathbb{P} dynamics are unrestricted to simplify estimation by the SR approach.

3.3 Model Estimation by the SR Approach

It is well known that DTSMs may be challenging to estimate, mainly because the parameters describing the market prices of risk can be hard to identify with highly persistent pricing factors. One may therefore encounter numerical instability and problems with local optima when estimating DTSMs (see, for instance, Duffee (2002)). Compared to the Gaussian ATSM, our model has potentially twice the number of parameters for the market prices of risk, suggesting that the proposed model may be quite demanding to estimate. To overcome this limitation and avoid problems with numerical instability, we draw on recent innovations in estimation methods for DTSMs, starting with the pioneering work of Joslin, Singleton and Zhu (2011). More specifically, we extend the SR approach of Andreasen and Christensen (2015) to estimate all parameters in our regime-dependent market price of risk by a modified OLS regression, even when allowing for measurement errors on all bond yields, as recommended by Hamilton and Wu (2014). In other words, all parameters in the market price of risk are obtained instantaneously within the SR approach, meaning that our extension of the Gaussian ATSM comes at no additional computational costs.¹²

We next describe the SR approach when adopted to our proposed DTSM with a regime-dependent market price of risk. In the interest of space, we only present the three steps in the SR approach and refer to Andreasen and Christensen (2015) for technical details and how to obtain standard errors.

Step 1: The first step of the SR approach estimates the latent pricing factors \mathbf{x}_t and the risk-neutral coefficients by a sequence of cross-section regressions. More formally, the risk-neutral coefficients are denoted by $\boldsymbol{\theta}_1 \equiv \begin{bmatrix} \boldsymbol{\theta}_{11}' & \boldsymbol{\theta}_{12}' \end{bmatrix}'$, where $\boldsymbol{\theta}_{11} \equiv \begin{bmatrix} \alpha & vec(\boldsymbol{\Phi})' \end{bmatrix}'$ and $\boldsymbol{\theta}_{12} \equiv vech(\boldsymbol{\Sigma}_{\mathbf{x}})$. Selecting

¹¹There exist other normalization schemes, for instance the one recently suggested by Joslin, Singleton and Zhu (2011).

¹²If some linear combinations of bonds yields are assumed to be perfectly priced by the model, the linear regression methods of Joslin, Singleton and Zhu (2011) and Hamilton and Wu (2012) may also be modified to estimate our regime-switching model with no additional computational costs compared to the Gaussian ATSM. However, we prefer the adopted approach because it also enables us to estimate the shadow rate extension of our regime-switching model considered below in Section 5.1, whereas the estimators of Joslin, Singleton and Zhu (2011) and Hamilton and Wu (2012) do not apply to nonlinear DTSMs.

 n_y maturities along the yield curve and accounting for measurement errors $v_{t,k}$, we then express (9) in stacked form as

$$\mathbf{y}_{t} = \tilde{\mathbf{A}}_{k}\left(\boldsymbol{\theta}_{1}\right) + \tilde{\mathbf{B}}_{k}\left(\boldsymbol{\theta}_{1}\right)\mathbf{x}_{t} + \mathbf{v}_{t},\tag{13}$$

where $\mathbf{y}_{t} \equiv \begin{bmatrix} y_{t,1} & y_{t,2} & \dots & y_{t,n_{y}} \end{bmatrix}'$ and similarly for $\tilde{\mathbf{A}}_{k}(\boldsymbol{\theta}_{1}), \tilde{\mathbf{B}}_{k}(\boldsymbol{\theta}_{1}),$ and \mathbf{v}_{t} .

When n_y is large relative to the number of pricing factors, the SR approach estimates \mathbf{x}_t by minimizing the squared distance between observed and model-implied bond yields, i.e. by regressing $\mathbf{y}_t - \mathbf{\tilde{A}}_k(\boldsymbol{\theta}_1)$ on $\mathbf{\tilde{B}}_k(\boldsymbol{\theta}_1)$. For our ATSM in (13), this implies

$$\hat{\mathbf{x}}_{t}\left(\boldsymbol{\theta}_{1}\right) = \left(\tilde{\mathbf{B}}_{k}\left(\boldsymbol{\theta}_{1}\right)'\tilde{\mathbf{B}}_{k}\left(\boldsymbol{\theta}_{1}\right)\right)^{-1}\tilde{\mathbf{B}}_{k}\left(\boldsymbol{\theta}_{1}\right)'\left(\mathbf{y}_{t}-\tilde{\mathbf{A}}_{k}\left(\boldsymbol{\theta}_{1}\right)\right)$$
(14)

for t = 1, 2, ..., T. The estimated factors are denoted $\{\hat{\mathbf{x}}_t (\boldsymbol{\theta}_1)\}_{t=1}^T$ because they are computed for a given $\boldsymbol{\theta}_1$. We then estimate $\boldsymbol{\theta}_1$ by pooling all squared residuals from these cross-section regressions and minimizing their sum with respect to $\boldsymbol{\theta}_1$, i.e.

$$\hat{\boldsymbol{\theta}}_{1}^{step1} = \arg\min_{\boldsymbol{\theta}_{1}\in\Theta_{1}} \frac{1}{Tn_{y}} \sum_{t=1}^{T} \left\| \mathbf{y}_{t} - \tilde{\mathbf{A}}_{k}\left(\boldsymbol{\theta}_{1}\right) - \tilde{\mathbf{B}}_{k}\left(\boldsymbol{\theta}_{1}\right) \hat{\mathbf{x}}_{t}\left(\boldsymbol{\theta}_{1}\right) \right\|^{2},$$

where Θ_1 denotes the feasible domain of $\boldsymbol{\theta}_1$ and $\|\mathbf{a}\| \equiv \sqrt{\sum_{i=1}^n a_i^2}$ for any $\mathbf{a} \in \mathbb{R}^n$.

Step 2: The second step estimates the \mathbb{P} dynamics of z_t and \mathbf{x}_t . To describe the procedure, let

$$\boldsymbol{\theta}_{2}^{\mathbf{x}} \equiv \left[\left(\boldsymbol{\lambda}_{0}^{(1)} \right)' \left(\boldsymbol{\lambda}_{0}^{(2)} \right)' \operatorname{vec} \left(\boldsymbol{\lambda}_{\mathbf{x}}^{(1)} \right)' \operatorname{vec} \left(\boldsymbol{\lambda}_{\mathbf{x}}^{(2)} \right)' \operatorname{vech} \left(\boldsymbol{\Sigma}_{\mathbf{x}} \right)' \right]'$$

and $\theta_2^{\mathbf{z}} \equiv \begin{bmatrix} \gamma_0 & \gamma_z & \gamma_x' & \Sigma_{zz} \end{bmatrix}'$ contain all the parameters governing the \mathbb{P} dynamics of \mathbf{x}_t and \mathbf{z}_t , respectively. Replacing the unobserved \mathbf{x}_t in (11) by $\hat{\mathbf{x}}_t \left(\hat{\theta}_1^{step1} \right)$ from the first step, we then estimate $\theta_2^{\mathbf{x}}$ by extending the SR approach of Andreasen and Christensen (2015) to \mathbb{P} dynamics with regime-switching. That is, we run a modified regression based on (11) that accounts for estimation uncertainty in $\hat{\mathbf{x}}_t \left(\hat{\theta}_1^{step1} \right)$, as described in Appendix A.5. The elements in $\theta_2^{\mathbf{z}}$ are obtained in a similar fashion based on (12) using the regression provided in Andreasen and Christensen (2015). Importantly, both $\left(\hat{\theta}_2^{\mathbf{x}} \right)^{step2}$ and $\left(\hat{\theta}_2^{\mathbf{z}} \right)^{step2}$ are given in closed form, meaning that all of the parameters in the \mathbb{P} distribution of z_t and \mathbf{x}_t are obtained instantaneously, including all coefficients in the market price of risk.

To ensure stationarity of $y_{t,k}$ we require \mathbf{x}_t to be stationary under the \mathbb{P} measure. This condition

holds if the loading matrices on \mathbf{x}_t in (11) are stable in recessions and in expansions, i.e. if all eigenvalues of the matrices $\mathbf{I} - \mathbf{\Phi} + \boldsymbol{\lambda}_{\mathbf{x}}^{(i)}$ for $i = \{1, 2\}$ are inside the unit circle.¹³ If one of these conditions are not satisfied, then we downscale $\mathbf{I} - \mathbf{\Phi} + \boldsymbol{\lambda}_{\mathbf{x}}^{(i)}$ by δ_i for $i = \{1, 2\}$ using the data-driven procedure of Andreasen and Meldrum (2014), which we describe in Appendix A.6.

Step 3: The matrix $\Sigma_{\mathbf{x}}$ is estimated in both the first and second step. As noted by Andreasen and Christensen (2015), $\hat{\Sigma}_{\mathbf{x}}^{step1}$ is estimated very inaccurately compared to $\hat{\Sigma}_{\mathbf{x}}^{step2}$, which is therefore the preferred estimate.¹⁴ Given this more efficient estimate of $\Sigma_{\mathbf{x}}$, we then condition on the value of $\hat{\Sigma}_{\mathbf{x}}^{step2}$ and re-estimate θ_{11} , i.e.

$$\hat{\boldsymbol{\theta}}_{11}^{step3} = \arg\min_{\boldsymbol{\theta}_{11}\in\Theta_{11}} \ \frac{1}{Tn_y} \sum_{t=1}^{T} \left\| \mathbf{y}_t - \tilde{\mathbf{A}}_k \left(\boldsymbol{\theta}_{11}; \hat{\boldsymbol{\Sigma}}_{\mathbf{x}}^{step2} \right) - \tilde{\mathbf{B}}_k \left(\boldsymbol{\theta}_{11}; \hat{\boldsymbol{\Sigma}}_{\mathbf{x}}^{step2} \right) \hat{\mathbf{x}}_t \left(\boldsymbol{\theta}_{11} \right) \right\|^2.$$

Given the estimated factors $\left\{ \hat{\mathbf{x}}_t \left(\hat{\boldsymbol{\theta}}_{11}^{step3}, \hat{\boldsymbol{\Sigma}}^{step2} \right) \right\}_{t=1}^T$, we finally update our estimates of $\boldsymbol{\theta}_2^{\mathbf{x}}$ and $\boldsymbol{\theta}_2^{\mathbf{z}}$ by re-running step 2.

3.4 Comparing to Existing DTSMs with Regime-Switching

When formulating DTSMs with regime-switching there is an inherent trade-off between the richness of a given model and the computational complexity related to bond pricing and estimation. We have therefore chosen to consider the most parsimonious model capable of matching the asymmetric properties of bond yields documented in Section 2. This implies that we omit regime-switching in $\Sigma_{\mathbf{x}}$ as in Ang, Bekaert and Wei (2008) or variation in $\Sigma_{\mathbf{x}}$ within regimes as in Bansal and Zhou (2002), because time-varying second moments are not required to reproduce the new properties of bond risk premia provided in Section 2.¹⁵ We also restrict the flexibility of our model by having observed regimes of either expansions or recessions, whereas most DTSMs with regime-switching consider unobserved regimes, although their estimated values often are closely related to the business cycle as in Bansal and Zhou (2002) and Dai, Singleton and Yang (2007). Similar to these two papers, the market price of factor risk in our model is allowed to change freely across regimes, whereas Ang, Bekaert and Wei (2008) consider a somewhat more restricted formulation. As in Dai, Singleton and Yang (2007), we also accommodate time-varying transition probabilities between regimes under the

¹³A formal proof of this result is provided in Theorem 6.12 of Pötscher and Prucha (1997).

¹⁴See Andreasen and Christensen (2015) for how to combine the two estimates in an optimal way.

¹⁵Furthermore, a direct extension of our estimator for $\Sigma_{\mathbf{x}}$ in (19) to accommodate regime-switching reveals only minor changes in the estimated value of $\Sigma_{\mathbf{x}}$ between recessions and expansions.

 \mathbb{P} measure when $\gamma_z \neq 0$ or $\gamma_x \neq 0$, whereas these probabilities are constant in Bansal and Zhou (2002) and Ang, Bekaert and Wei (2008).

4 Empirical Findings

This section estimates our DTSM with regime-switching in the case of three pricing factors $(n_x = 3)$, given a total of four state variables when including z_t . Here, we use the same Fama-Bliss dataset as applied in Section 2. We first discuss the estimated coefficients in Section 4.1, before studying the ability of our regime-switching model to match the observed asymmetry in the U.S. yield curve in Section 4.2. The implied estimates of excess returns and term premia from our regime switching model and the Gaussian ATSM are finally compared in Section 4.3.

4.1 Estimation Results

Our proposed ATSM with regime switching in the market price of risk is indexed by $\mathcal{M}_{\lambda_0,\lambda_x}^{ATSM}$, where subscripts indicate whether λ_0 and/or λ_x are allowed to switch between expansions and recessions. Table 3 shows that the first factor is very persistent under the \mathbb{Q} measure with $\Phi(1,1)\approx 1.2\times 10^{-8}$ in all models and may therefore be interpreted as a level factor. The low value of $\Phi(1,1)$ implies that the intercept in the short rate is unidentified and we therefore let $\alpha = 0$. The second and the third factor display less persistence under \mathbb{Q} in all considered models and may therefore be interpreted as a slow and fast decaying factors. Unlike $\hat{\Phi}$, the estimates of Σ_x are somewhat affected by regime switching in the market price of risk, although the sign of the off-diagonal elements in $\hat{\Sigma}_x$ is similar across all models. Table 3 also provides the estimated dynamics for economic activity, which displays moderate persistence with $\gamma_z = 0.93$ across all specifications of λ_t . We also find that each of the pricing factors has a negative effect on economic activity. Importantly, the null hypothesis that $\gamma_x (1,1) = \gamma_x (2,1) = \gamma_x (3,1)$ is clearly rejected using a Wald test with a *p*-value of 0.000 for all models, implying that the policy rate is not a sufficient statistic for economic activity according to our model.

< Table 3 about here >

The estimates of the market price of risk are provided in Table 4. For the Gaussian ATSM \mathcal{M}^{ATSM} all elements of λ_0 are significantly different from zero, meaning that investors on average require compensation for exposure to each of the pricing factors. We also find the familiar result that time-variation in level risk through $\lambda_{\mathbf{x}}(1,:)$, $\lambda_{\mathbf{x}}(2,1)$, and $\lambda_{\mathbf{x}}(3,1)$ is significant and controls much of the variability in the market price of risk and hence excess returns (see, for instance, Cochrane and Piazzesi (2008)).

We next allow the constant λ_0 in the market price of risk to switch in $\mathcal{M}_{\lambda_0}^{ATSM}$. During recessions $\lambda_0(2, 1)$ has a higher value than in expansions, whereas the opposite holds for $\lambda_0(3, 1)$. These differences are significant as we reject the null hypothesis that $\lambda_0^{(1)} = \lambda_0^{(2)}$ using a Wald test (*p*-value of 0.0124). Suppose instead that only λ_x is allowed to switch in the market price of risk, i.e. λ_0 is constant as in the Gaussian ATSM. For $\mathcal{M}_{\lambda_x}^{ATSM}$ in Table 4 we find that time-variation in the second factor through $\lambda_x(2,2)$ now carries a negative price, although most predominately in recessions, whereas variation in this second factor is priced positively in \mathcal{M}^{ATSM} and $\mathcal{M}_{\lambda_0}^{ATSM}$. We also observe that $\lambda_x(3,2)$ and $\lambda_x(3,3)$ have opposite signs in recessions and expansions. Despite the somewhat wide standard errors for $\lambda_x^{(2)}$, the null hypothesis that $\lambda_x^{(1)} = \lambda_x^{(2)}$ is clearly rejected (*p*-value of 0.0000).

We finally consider the full model $\mathcal{M}_{\lambda_0,\lambda_x}^{ATSM}$, where both λ_0 and λ_x are allowed to switch. We once again find that the value of λ_0 (2, 1) changes between recessions and expansions, and that time-variation in the second factor through λ_x (2, 2) carries a positive price in expansions but a negative price in recessions. The latter implies that the pricing factors and hence interest rates are more persistent in expansions than in recessions when measured by the largest eigenvalue of $\mathbf{I} - \mathbf{\Phi} + \lambda_x^{(i)}$, which is 0.9957 in expansions and 0.9605 in recessions. This finding is thus similar to Bansal and Zhou (2002). However, most elements in $\lambda_0^{(2)}$ and $\lambda_x^{(2)}$ for the recession regime are estimated somewhat imprecisely, and we are therefore unable to reject the null hypothesis of no regime switching in λ_t (*p*-value of 0.26), although we rejected it for both $\mathcal{M}_{\lambda_0}^{ATSM}$ and $\mathcal{M}_{\lambda_x}^{ATSM}$. This indicates that our general specification for regime-switching in λ_0 and λ_x may be simplified without affecting the model's ability to fit the data. We return to this possibility below in Section 5.2.

< Table 4 about here >

4.2Explaining Asymmetries in the Yield Curve

We next examine whether the full model with switching in λ_0 and λ_x can match the asymmetries in the yield curve from Section 2. The model-implied moments are here obtained by running the regressions in (1) to (4) on a simulated sample of 100,000 observations using the estimates of $\mathcal{M}_{\lambda_0,\lambda_x}^{ATSM}$ reported above.

The top part of Figure 4 shows that $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ convincingly matches both intercepts and slope coefficients in ordinary Campbell-Shiller regressions and thus preserves the ability of the Gaussian ATSM in matching these unconditional properties of bond risk premia. The next question is whether $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ meets the challenge of replicating the switch in the Campbell-Shiller regressions across expansions and recessions. The bottom part of Figure 4 shows that $\mathcal{M}_{\lambda_0\lambda_T}^{ATSM}$ reproduces the negative slope estimates in expansions and the *positive* slope coefficients in recessions, although the latter are somewhat higher than the empirical moments. In addition, $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ matches almost perfectly the switch in intercepts.

We next explore whether $\mathcal{M}_{\lambda_0\lambda_n}^{ATSM}$ also replicates the predictability in bond returns implied by i) the yield spread, *ii*) the forward spread, and *iii*) the CP factor. The left part of Figure 5 shows that $\mathcal{M}_{\lambda_0\lambda_m}^{ATSM}$ matches the slope coefficients from ordinary return regressions using each of these predictors. The ability of our model to reproduce the predictability of the CP factor suggests that the behavior of this factor to a large extent is spanned by the three canonical pricing factors in our model.¹⁶ The right part of Figure 5 further shows that $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ also generates much of the observed asymmetry in return predictability across business cycles for all three predictors. That is, our model implies strong predictability of bond returns during expansions, whereas recessions are characterized by weak predictability and often with the 'wrong' sign.

< Figure 4 and 5 about here >

Combining the results from Figure 4 and 5, we conclude that our regime switching model $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ to a large extent is capable of explaining the asymmetric behavior of bond yields from Section 2^{17} Thus, the switch in the Campbell-Shiller and return regressions across expansions and recessions

¹⁶ A similar finding is reported in Dai, Singleton and Yang (2004). ¹⁷ We have also analyzed the ability of $\mathcal{M}_{\lambda_0}^{ATSM}$ and $\mathcal{M}_{\lambda_x}^{ATSM}$ to match the observed asymmetric patterns in the data. $\mathcal{M}_{\lambda_0}^{ATSM}$ matches the sign switch in intercepts but not in the slope coefficients of our modified Campbell-Shiller regression and it fails to generate low return predictability in recessions. $\mathcal{M}_{\lambda_x}^{ATSM}$ does not match neither intercepts nor slope coefficients in the modified Campbell-Shiller regressions and it generates too high return predictability in expansions.

may be rationalized by a re-pricing of risk among bond investors when the U.S. economy enters recessions.

4.3 Model-implied Excess Returns and Term Premia

The presence of regime-switching in the market price of risk has profound implications for bond risk premia. Figure 6 therefore plots expected excess returns from $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ and \mathcal{M}^{ATSM} at the 10-year maturity, with NBER recessions indicated by the shaded regions. The two models provide very similar dynamics for expected excess returns in expansions. However, before and at the start of recessions, we generally find that expected excess returns in $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ increase more than in \mathcal{M}^{ATSM} . The prime examples are the two recessions in the early 1980s. This effect arises because $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ assigns an increasing fraction of the forecast distribution to the recession regime, where all bond yields are predicted to be substantially lower than in expansions. This in turn raises bond prices and hence expected excess returns. This 'recession amplifier' disappears towards the middle and the end of recessions where excess returns fall, after which excess returns in $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ approach the level implied by \mathcal{M}^{ATSM} .

< Figure 6 about here >

Any estimate of bond risk premia is model-dependent, and it may therefore in general be challenging to argue in favor of one set of estimates compared to another. However, the regression evidence we provide in Figures 4 and 5 clearly indicates that our regime-switching model matches important properties of bond returns unlike the Gaussian ATSM, suggesting that $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ gives more accurate estimates of bond risk premia compared to \mathcal{M}^{ATSM} . Another way to illustrate this is to run Mincer-Zarnowitz regressions of realized excess returns on model-implied expected excess returns $\mathbb{E}_t [xhpr_{t+m,k}]$. Conditioning on the business cycle as in Section 2, we consider

$$xhpr_{t+m,k} = \alpha_{0,k}^{EXP} \left(1 - 1_{\{NBER\}} \right) + \alpha_{1,k}^{EXP} \left(1 - 1_{\{NBER\}} \right) \mathbb{E}_t \left[xhpr_{t+m,k} \right]$$
(15)
+ $\alpha_{0,k}^{REC} 1_{\{NBER\}} + \alpha_{1,k}^{REC} 1_{\{NBER\}} \mathbb{E}_t \left[xhpr_{t+m,k} \right] + u_{t,k},$

with m = 3, where recessions and expansions are identified using the NBER recessions dates $1_{\{NBER\}}$ for comparability with Figure 6. The slope coefficients $\alpha_{1,k}^{EXP}$ and $\alpha_{1,k}^{REC}$ are reported in Table 5 when pooling excess returns in bins of two years along the maturity range. Consistent with

the regression evidence in Figure 4 and 5, \mathcal{M}^{ATSM} performs well in expansions with $\alpha_{1,k}^{EXP}$ close to the desired value of one, but the model struggles during recessions, where $\alpha_{1,k}^{REC}$ is substantially below one and even negative for bond yields with six or more years to maturity. That is, the Gaussian ATSM predicts excess returns for medium- and long-term bonds in the wrong direction during recessions. Our regime switching model largely alleviates this shortcoming with both $\alpha_{1,k}^{EXP}$ and $\alpha_{1,k}^{REC}$ in the range of 0.75 to 0.54, meaning that $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ predicts excess returns of all bond yields in the right direction during expansions and recessions.

< Table 5 about here >

The lower part of Figure 6 reports model-implied term premia, defined as $TP_{t,k} \equiv y_{t,k} - \frac{1}{k} \sum_{i=0}^{k-1} \mathbb{E}_t [r_{t+i}]$, which is another commonly used measure of bond risk premia. In expansions, we find term premia to be very similar for \mathcal{M}^{ATSM} and $\mathcal{M}^{ATSM}_{\lambda_0\lambda_x}$, but substantial differences appear before and at the start of recessions, where term premia in $\mathcal{M}^{ATSM}_{\lambda_0\lambda_x}$ tend to increase more than in \mathcal{M}^{ATSM} . That is, term premia is more counter-cyclical in our regime-switching model compared to the standard Gaussian ATSM.

5 Additional Analysis

This section studies in greater detail the ability of regime-switching in the market price of risk to explain the documented asymmetries in the U.S. yield curve from Section 2. We first consider the implications of extending $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ to enforce the zero lower bound (ZLB) in Section 5.1. Given our findings in Section 4.1, we then present a simplified version of $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ in Section 5.2, which we use to provide an economic explanation for the observed switch in the Campbell-Shiller and return regressions across expansions and recessions. The ability of DTSMs with linear unspanned and spanned macroeconomic risk to match loadings from our modified Campbell-Shiller regressions are finally explored in Section 5.3.

5.1 Accounting for the Zero Lower Bound

An important feature of our regime-switching model is to generate a larger fall in expected bond yields just before and during recessions than implied by the Gaussian ATSM. Given that short rates typically are low during recessions and may be constrained by the ZLB, it seems natural to explore whether the better performance of our model is robust to enforcing the ZLB. We address this question by presenting a shadow rate extension of our regime-switching model where the short rate is given by $r_t = \max \{0, \alpha + \beta' \mathbf{x}_t\}$, but the model is otherwise identical to the one described in Section 3.2. Hence, the proposed shadow rate model (SRM) has Gaussian pricing factors under the \mathbb{Q} measure and bond prices may therefore be computed by the second-order approximation advocated in Priebsch (2013) when formulated in discrete time. This SRM with regime-switching $\mathcal{M}_{\lambda_0\lambda_x}^{SRM}$ is estimated as described in Section 3.3, except (14) is replaced by nonlinear cross-section regressions to extract the pricing factors. The estimates are provided in Tables 6 and 7.

Figure 7 shows that $\mathcal{M}_{\lambda_0\lambda_x}^{SRM}$ provides an extremely close fit to intercepts in ordinary Campbell-Shiller regressions and even improves upon the slope coefficients in these regressions for mediumand long-term bond yields compared to $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$. This improvement is further seen to carry over to the slope coefficients during expansions, whereas $\mathcal{M}_{\lambda_0\lambda_x}^{SRM}$ largely generates the same intercepts and slope coefficients for the recession regime as found for $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$. Although not reported below, we also find that $\mathcal{M}_{\lambda_0\lambda_x}^{SRM}$ matches the return regressions just as well as seen for $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$. Finally, $\mathcal{M}_{\lambda_0\lambda_x}^{SRM}$ also forecasts realized excess returns in the correct direction during both expansions and recessions, as shown in Table 5.

$$<$$
 Table 6-7, Figure 7 about here $>$

Thus, our proposed explanation for the documented asymmetries in the U.S. yield curve, i.e. that investors re-price risk during recessions, is robust to accounting for the ZLB.

5.2 A Simplified Regime-Switching Model

To provide a simplified version of $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ with a clear economic interpretation, it seems beneficial to first impose more structure on the factor loadings than implied by our fully flexible regime-switching model $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$. We therefore first note that the null hypothesis of $\Phi(1,1) = 0$ and $\Phi(2,2) = \Phi(3,3)$ is not rejected using a Wald test (*p*-value of 0.6406). Given these restrictions, it is straightforward to show that the factor loadings in $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ simplify to those in the no-arbitrage Nelson-Siegel (AFNS) model of Christensen, Diebold and Rudebusch (2011) when using its discrete-time formulation in Fontaine and Garcia (2012). That is, we let $\boldsymbol{\beta} = \begin{bmatrix} 1 & \frac{1-e^{-\tilde{\lambda}}}{\tilde{\lambda}} & \frac{1-e^{-\tilde{\lambda}}}{\tilde{\lambda}} - e^{-\tilde{\lambda}} \end{bmatrix}'$ and impose $\Phi(2,2) = \Phi(3,3) = 1 - e^{-\tilde{\lambda}}$ with $\Phi(2,3) = -\tilde{\lambda}e^{-\tilde{\lambda}}$, whereas all the remaining elements

of Φ are zero. Given this specification, the pricing factors now have the familiar interpretation as representing the level, slope, and curvature of the yield curve.

The results from Section 4.1 indicate that not all elements in λ_0 and λ_x switch between expansions and recessions, and we therefore search for the most parsimonious specification of λ_t capable of matching the switch in the Campbell-Shiller regressions across the business cycle. Inspired by the return regressions with the yield spread in Section 2.2, the proposed model displays only switching in λ_0 (2, 1) and λ_x (2, 2), both related to the slope factor at position two in our state vector. This restricted model is denoted $\mathcal{M}^{AFNS}_{\lambda_0(2,1)\lambda_x(2,2)}$ and has only two additional coefficients in the market price of risk compared to the Gaussian ATSM. The top part of Figure 8 shows that $\mathcal{M}^{AFNS}_{\lambda_0(2,1)\lambda_x(2,2)}$ captures the overall pattern in intercepts and slope coefficients from ordinary Campbell-Shiller regressions. More surprising, perhaps, is the ability of this very parsimonious regime-switching specification to also capture the switch in both intercepts and slope coefficients in the Campbell-Shiller regressions between expansions and recessions. Thus, simply allowing for regime-switching in the dynamics of the slope factor is sufficient to reproduce the loadings in our modified Campbell-Shiller regressions.

To see what drives these results for $\mathcal{M}_{\lambda_0(2,1)\lambda_x(2,2)}^{AFNS}$, we first note from Table 7 that $\mathcal{M}_{\lambda_0(2,1)\lambda_x(2,2)}^{AFNS}$ has a lower value of $\lambda_0(2,1)$ in recessions than in expansions and that $\lambda_x(2,2)$ switches sign between the two regimes. The latter arises because $1 - \Phi(2,2) + \lambda_x^{(1)}(2,2) = 0.97$ in expansions, whereas the corresponding estimate in recessions is 0.73. Given that $\Phi(2,2) = 0.0381$ in $\mathcal{M}_{\lambda_0(2,1)\lambda_x(2,2)}^{AFNS}$, this change in the persistence of the slope factor then generates a switch in the market price of risk from a insignificant positive value of $\lambda_x^{(1)}(2,2) = 0.0081$ in expansions to a significant negative value of $\lambda_x^{(2)}(2,2) = -0.2363$ in recessions.

To analyze the effects of these estimates for quarterly excess returns, as considered in our implementation of the Campbell-Shiller regressions in (1) and (2), it is useful to first understand their impact on monthly excess returns which are available in closed form and proportional to $\mathbf{B}'_{k-1}\boldsymbol{\lambda}_t$.¹⁸ Given that $B_{k-1}(2,1) < 0$, a more negative value of $\lambda_0(2,1)$ in recessions therefore increases monthly excess returns, as investors require a larger compensation for slope risk in this regime. On the other hand, a negative value of $\lambda_x^{(2)}(2,2)$ in recessions has a positive effect on $\boldsymbol{\lambda}_{\mathbf{x}}^{(2)}\mathbf{x}_t$ with an upward sloping yield curve, because the slope factor with the AFNS loadings is defined as the short rate *minus* the long rate (see Diebold and Li (2006)), meaning that $\lambda_x^{(2)}(2,2)$ in total

 $^{^{18}}$ Here, we ignore a minor convexity term in the expression for monthly excess returns. The exact expression is provided in our Online Appendix.

has a negative impact on monthly excess returns when accounting for $B_{k-1}(2,1) < 0$. Figure 9 shows that these partial effects carry over to the quarterly excess return for a 10-year bond which we compute by Monte Carlo integration. Note also how fears of recessions on its own may affect expected quarterly excess returns (as in the mid 1990s). We finally emphasize that $\mathcal{M}_{\lambda_0(2,1)\lambda_x(2,2)}^{AFNS}$ predicts realized excess returns with the right sign during expansions and recessions according to the Mincer-Zarnowitz regressions in Table 5, despite $\mathcal{M}_{\lambda_0(2,1)\lambda_x(2,2)}^{AFNS}$ only having two parameters that switch between regimes.

Thus, the narrative implied by our model is as follows. When the U.S. economy enters a recession, bond investors immediately re-price the risks attached to economic activity with a switch in $\lambda_0(2,1)$ and therefore require a higher compensation for exposure to the slope of the yield curve. At the start of a recession, this effect generally dominates the one from a switch in $\lambda_x(2,2)$ according to Figure 9, as the yield curve here tends to be fairly flat, and this increases expected excess returns. However, as monetary policy becomes more accommodating during the course of a recession, the lower short rate generates a steepening of the yield curve and reduces the risks attached to low future economic activity. This in turn strengthens the effect from the switch in $\lambda_x(2,2)$, which generally has a negative impact on excess returns according to Figure 9. Thus, our model suggests that the Federal Reserve is able to remove some of the risks attached to recessions, as accommodating monetary policy reduces the required risk compensation in the bond market. Importantly, the switch in $\lambda_x(2,2)$ from positive in expansions to negative in recessions implies that excess returns and the short rate both *fall* during this phase of a recession and hence become positively correlated, as required to match the positive slope coefficients in the Campbell-Shiller regressions during recessions. The Gaussian ATSM does not allow for a switch in $\lambda_x(2,2)$, meaning that this model does *not* imply that accommodating monetary policy is able to remove some of the risks in the bond market during recessions. As a result, excess returns tend to increase throughout recessions in the Gaussian ATSM, as also seen in Figure 6. After a recession, $\mathcal{M}_{\lambda_0(2,1)\lambda_x(2,2)}^{AFNS}$ reverts back to the familiar setting described in Dai and Singleton (2002), where excess returns fall when the short rate increases, and vice versa. That is, excess returns and the short rate are once again negatively correlated as required to match the negative slope in the Campbell-Shiller regressions during expansions.

< Figure 8 about here >

5.3 Models with Linear Unspanned and Spanned Macroeconomic Risk

Our proposed mechanism to explain the documented asymmetries in U.S. bond yields modifies the essentially affine specification for the market price of risk such that it only holds locally for expansions and recessions. Hence, λ_t depends nonlinearly on the business cycle, as captured by the state variable z_t for PMI. To explore whether this nonlinearity is necessary to match the switch in the Campbell-Shiller regressions, we next follow Joslin, Priebsch and Singleton (2014) and briefly study the case where λ_t only depends linearly on z_t . That is, we now let the market price of risk be essentially affine in all four state variables. As above, z_t remains unspanned by the current yield curve, i.e. $\beta(4,1) = 0$, given our specification with z_t appearing last in the state vector. The estimates for this model \mathcal{M}^{Linear} are provided in Tables 6 and 7.

Figure 10 shows that \mathcal{M}^{Linear} provides a very close fit to both intercepts and slope coefficients in ordinary Campbell-Shiller regressions. This is in line with previous findings in the literature, as this model nests the standard three-factor Gaussian ATSM. The lower part of Figure 10 shows that \mathcal{M}^{Linear} also matches the regime-dependent intercepts in the Campbell-Shiller regressions but only the negative slope coefficients in expansions. That is, \mathcal{M}^{Linear} cannot generate positive slope coefficients during recessions. Table 5 further shows that \mathcal{M}^{Linear} only has predictive power for realized excess returns in expansions but not in recessions. Hence, it must be the nonlinear effect of z_t on the market price of risk that allows $\mathcal{M}^{ATSM}_{\lambda_0\lambda_x}$, $\mathcal{M}^{SRM}_{\lambda_0\lambda_x}$, and $\mathcal{M}^{AFNS}_{\lambda_0(2,1)\lambda_x(2,2)}$ to match the slope coefficients in our modified Campbell-Shiller regression and predict excess returns with the right sign during expansions and recessions.

Following the work of Bauer and Rudebusch (2016), we finally explore whether these results for \mathcal{M}^{Linear} are robust to letting z_t be spanned by the current yield curve. That is, we once again let the market price of risk be essentially affine in all four state variables but omit the zero restriction for β (4, 1). Figure 10 shows that this model $\mathcal{M}^{Spanned}$ displays broadly the same performance as \mathcal{M}^{Linear} in terms of matching intercepts and slope coefficients from ordinary and modified Campbell-Shiller regressions. As a result, the Mincer-Zarnowitz regressions for excess returns in Table 5 are also very similar for the two models. Hence, the spanning assumption of macroeconomic risk, as measured by the PMI, does not seem essential for the aspects of U.S. bond yields studied in this paper.

6 Conclusion

By conditioning the classic Campbell-Shiller (1991) regressions on the business cycle, we identify a strong asymmetric pattern in the relation between yield spreads and subsequent yield changes. When the economy is expanding, we find the familiar pattern of mostly positive intercepts and negative regression slopes that decrease with maturity. However, when the economy is contracting, we observe negative intercepts and positive regression slopes that mostly increase with maturity, i.e. the complete opposite pattern. We also show that this asymmetric effect has profound implications for bond return predictability, as the classic yield-based predictions have strong forecasting power for excess returns in expansions but not in recessions.

To explain these new empirical findings, the Gaussian ATSM is extended with business cycle dependent loadings in the market price of risk. We show that this model reproduces the empirical intercepts and slope coefficients from ordinary Campbell-Shiller regressions and, in addition, matches evidence from our modified Campbell-Shiller regressions conditioning on the business cycle. We also show that our model predicts realized excess returns with the right sign during both expansions and recessions unlike the Gaussian ATSM, which predicts excess returns for medium- and long-term bonds in the wrong direction during recessions. Our model also replicates the observed asymmetry in return predictability, i.e. that the classic yield-based predictors have strong forecasting power in expansions but not in recessions. To account for these asymmetric patterns in the yield curve across the business cycle, our model generates a *negative* relation between the short rate and excess returns in expansions (as in the Gaussian ATSM), but a *positive* relation in recessions. Thus, our model suggests that the Federal Reserve is able to remove some of the risks attached to recessions, as accommodating monetary policy reduces the required risk compensation in the bond market. This effect of monetary policy is not present in the Gaussian ATSM, which therefore provides less accurate estimates of bond risk premia and expected future short rates compared to our regime-switching model. Accordingly, our model suggests that the positive slope coefficients in the Campbell-Shiller regressions emerge because accommodating monetary policy is able to eleminate some of the risks in the bond market during recessions.

Our simple, yet powerful, approach of conditioning the study of bond markets on the business cycle obviously goes beyond what is covered in the present paper. Ongoing research by Andreasen, Møller and Sander (2016) shows that bond markets in several other countries display a similar switch in the Campell-Shiller regressions as found in the present paper, both when conditioning on local recessions or U.S. recessions. Andreasen, Møller and Sander (2016) further show that this finding has important implications for exchange rate dynamics, as the slope coefficients in uncovered interest parity (UIP) regressions also switch between expansions and regressions. Thus, recessions may not only have profound effects for the local U.S. bond investor but also for other bond markets and related exchange rate markets.

A Appendix

A.1 Data

We use monthly bond yields from an unsmoothed Fama and Bliss (1987) dataset running from 1961:06-2013:12. We limit the analysis to maturities from m/12 to 10 years with m-month increments where m = 3. A similar panel of yields is used in Adrian, Crump and Moench (2013) to obtain their pricing factors. At the very long end of the yield curve, we do not observe all increments and, hence, we interpolate between bond yields of the two nearest maturities.¹⁹ We also report results for the Gürkaynak, Sack and Wright (2007) dataset, which is based on a static Svensson model for computing smooth yield curves.

A.2 Wald Tests

We use Wald tests to examine whether the coefficients in our modified Campbell-Shiller regressions in (2) change across expansions and recessions. In particular, we test the following joint hypotheses:

$$H_0: \alpha_k^{REC} = \alpha_k^{EXP}, \quad k = 6, 9, 12, ..., 120$$
(16)

$$H_0: \beta_k^{REC} = \beta_k^{EXP}, \quad k = 6, 9, 12, ..., 120$$
(17)

The tests are carried out by setting up the regressions in a GMM framework, where all parameters are collected in

$$oldsymbol{\Theta}' = \left[oldsymbol{ heta}_6', \ oldsymbol{ heta}_9', \ \ ..., \ oldsymbol{ heta}_{120}'
ight],$$

with $\boldsymbol{\theta}'_{k} = \left[\alpha_{k}^{EXP}, \ \beta_{k}^{EXP}, \ \alpha_{k}^{REC}, \ \beta_{k}^{REC}\right]$. The sample moments for the system of linear regressions are

$$\mathbf{g}_{T}(\mathbf{\Theta})' = \begin{bmatrix} \mathbf{g}_{T,6}', \mathbf{g}_{T,9}', ..., \mathbf{g}_{T,120}' \end{bmatrix},$$

where $\mathbf{g}_{T,k} = \frac{1}{T} \sum_{t=1}^{T} \mathbf{f}_{t,k}, \ \mathbf{f}_{t,k} = \tilde{u}_{t+m,k} \mathbf{z}_{t,k}$, and

$$\mathbf{z}_{t,k}' = \left[\mathbf{1}_{\{z_t \ge c\}}, \frac{m}{k-m} \mathbf{1}_{\{z_t \ge c\}} \left(y_{t,k} - y_{t,m}\right), \left(1 - \mathbf{1}_{\{z_t \ge c\}}\right), \frac{m}{k-m} \left(1 - \mathbf{1}_{\{z_t \ge c\}}\right) \left(y_{t,k} - y_{t,m}\right)\right].$$

Expressing our null hypothesis as $\mathbf{R}\Theta = \mathbf{c}$, a Wald test is then given by

$$W = \left(\mathbf{R}\widehat{\boldsymbol{\Theta}} - \mathbf{c}\right)' \left(\mathbf{R} \left(\mathbf{D}_{T}'\mathbf{S}_{T}^{-1}\mathbf{D}_{T}\right)^{-1} \mathbf{R}'/T\right)^{-1} \left(\mathbf{R}\widehat{\boldsymbol{\Theta}} - \mathbf{c}\right) \sim \chi_{K}^{2},$$

where \mathbf{D}_T is the Jacobian of the moment conditions and \mathbf{S}_T is the spectral density matrix obtained by the Newey-West estimator using m + 1 lags. For the hypothesis in (16), we have

$$\mathbf{R} = \mathbf{ID}_K \otimes [1, 0, -1, 0],$$

where \mathbf{ID}_K is the identity matrix with dimension K and $\mathbf{c} = \mathbf{0}$. K is the number of maturities we

¹⁹Fama-Bliss forward rates are known to be somewhat rough for longer maturities. We therefore smooth them using a 12-month equal-weighted window.

examine and is equal to 39 for the full panel. For the hypothesis in (17), we have

$$\mathbf{R} = \mathbf{ID}_K \otimes [0, 1, 0, -1],$$

and $\mathbf{c} = \mathbf{0}$.

A.3 Lynch-Wachter Estimator

Conducting the tests for the full sample period involves making some implementation choices as the Fama-Bliss panel is unbalanced. We use the adjusted moments estimator of Lynch and Wachter (2013), which is an efficient way of exploiting the long sample. For simplicity we partition the moments in two parts: One part where we have yields for full the period from 1961:06-2013:12 and one part where we only have yields starting from 1971:11. The first group contains moments on maturities up to 66 months, while the second group contains those with maturities from 69 to 120 months. In principle, we could make many partitions based on the available observations for each moment, but this will heavily complicate the computations. We refer to Lynch and Wachter (2013) and our Online Appendix for further details.

A.4 Bootstrap

We address potential small sample distortions in the inference by a non-parametric stationary bootstrap procedure in order to account for possible time series dependencies and to preserve any cross-sectional dependencies in the data. The bootstrap procedure resamples the data in blocks of consecutive observations of the left and right hand side of the Campbell-Shiller regressions. We determine the optimal average block size using the Politis and White (2004) approach. When drawing the blocks, we make sure to preserve cross-sectional correlations by resampling from the same time points for all maturities. For each of the 100,000 bootstrap samples, we estimate the coefficients in the Campbell-Shiller regressions.²⁰ Based on a matrix of the 100,000 simulated coefficient estimates, we can then compute the variances and covariances of the estimates across maturities.

A.5 Step 2 of the SR approach: Regime-Switching in the Time Series Regression

This subsection describes how to estimate (11) by GMM when accounting for measurement errors in the estimated pricing factors. For notational convenience, we express (11) as

$$\mathbf{x}_{t+1} = \mathbf{1}_{\{z_t \ge c\}} \mathbf{h}_0^{(1)} + \left(1 - \mathbf{1}_{\{z_t \ge c\}}\right) \mathbf{h}_0^{(2)} + \mathbf{h}_{\mathbf{x}}^{(1)} \mathbf{x}_t^{(1)} + \mathbf{h}_{\mathbf{x}}^{(2)} \mathbf{x}_t^{(2)} + \mathbf{w}_{t+1},$$

where $\mathbf{h}_{0}^{(i)} \equiv \boldsymbol{\lambda}_{0}^{(i)}, \mathbf{h}_{\mathbf{x}}^{(i)} \equiv \mathbf{I} - \boldsymbol{\Phi} + \boldsymbol{\lambda}_{\mathbf{x}}^{(i)}, \mathbf{x}_{t}^{(1)} \equiv \mathbf{1}_{\{z_{t} \geq c\}} \mathbf{x}_{t}, \mathbf{x}_{t}^{(2)} \equiv (1 - \mathbf{1}_{\{z_{t} \geq c\}}) \mathbf{x}_{t}, \text{ and } \mathbf{w}_{t+1} \equiv \boldsymbol{\Sigma}_{\mathbf{x}} \boldsymbol{\epsilon}_{t+1}$ for $i = \{1, 2\}$. Using the same procedure as in Andreasen and Christensen (2015), we estimate $\boldsymbol{\theta}_{2}^{\mathbf{x}}$

²⁰We require at least 15 recession observations in each simulated sample.

based on

$$\begin{bmatrix}
E \begin{bmatrix} \hat{\mathbf{w}}_{t+1} \mathbf{1}_{\{z_t \ge c\}} \end{bmatrix} \\
E \begin{bmatrix} \hat{\mathbf{w}}_{t+1} \left(1 - \mathbf{1}_{\{z_t \ge c\}} \right) \end{bmatrix} \\
E \begin{bmatrix} \hat{\mathbf{w}}_{t+1} \left(\hat{\mathbf{x}}_t^{(1)} \right)' \\
E \begin{bmatrix} \hat{\mathbf{w}}_{t+1} \left(\hat{\mathbf{x}}_t^{(2)} \right)' \end{bmatrix} \\
Var \left(\hat{\mathbf{w}}_{t+1} \right)
\end{bmatrix} = \begin{bmatrix} \mathbf{0} \\ \mathbf{0} \\
Cov \left(\mathbf{u}_{t+1}^{(1)}, \mathbf{u}_t^{(1)} \right) - \mathbf{h}_{\mathbf{x}}^{(1)} Var \left(\mathbf{u}_t^{(1)} \right) \\
Cov \left(\mathbf{u}_{t+1}^{(2)}, \mathbf{u}_t^{(2)} \right) - \mathbf{h}_{\mathbf{x}}^{(2)} Var \left(\mathbf{u}_t^{(2)} \right) \\
Var \left(\mathbf{w}_{t+1} \right) + \mathbf{\Omega}_{t+1}
\end{bmatrix}, \quad (18)$$

where

$$\begin{split} \mathbf{\Omega}_{t+1} &\equiv Var\left(\mathbf{u}_{t+1}\right) + \mathbf{h}_{\mathbf{x}}^{(1)}Var\left(\mathbf{u}_{t}^{(1)}\right) \left(\mathbf{h}_{\mathbf{x}}^{(1)}\right)' + \mathbf{h}_{\mathbf{x}}^{(2)}Var\left(\mathbf{u}_{t}^{(2)}\right) \left(\mathbf{h}_{\mathbf{x}}^{(2)}\right)' \\ &- Cov\left(\mathbf{u}_{t+1}^{(1)}, \mathbf{u}_{t}^{(1)}\right) \left(\mathbf{h}_{\mathbf{x}}^{(1)}\right)' - \mathbf{h}_{\mathbf{x}}^{(1)}Cov\left(\mathbf{u}_{t}^{(1)}, \mathbf{u}_{t+1}^{(1)}\right) \\ &- Cov\left(\mathbf{u}_{t+1}^{(2)}, \mathbf{u}_{t}^{(2)}\right) \left(\mathbf{h}_{\mathbf{x}}^{(2)}\right)' - \mathbf{h}_{\mathbf{x}}^{(2)}Cov\left(\mathbf{u}_{t}^{(2)}, \mathbf{u}_{t+1}^{(2)}\right). \end{split}$$

Here, \mathbf{u}_t refers to the estimation uncertainty in the estimated pricing factors, i.e. $\hat{\mathbf{x}}_t = \mathbf{x}_t + \mathbf{u}_t$, where $\mathbf{u}_t^{(1)} \equiv \mathbf{1}_{\{z_t \geq c\}} \mathbf{u}_t$ and $\mathbf{u}_t^{(2)} \equiv (1 - \mathbf{1}_{\{z_t \geq c\}}) \mathbf{u}_t$. As in Andreasen and Christensen (2015), all required moments of \mathbf{u}_t follow from the first step of the SR approach. The solution to the first four moments conditions in (18) is given in closed form by

$$\begin{bmatrix} \widehat{\mathbf{h}}_{0}^{(1)} & \widehat{\mathbf{h}}_{0}^{(2)} & \widehat{\mathbf{h}}_{\mathbf{x}}^{(1)} & \widehat{\mathbf{h}}_{\mathbf{x}}^{(2)} \end{bmatrix} = \left(\sum_{t=1}^{T-1} \widehat{\mathbf{x}}_{t+1} \mathbf{a}_{t}' - \sum_{t=1}^{T-1} \widehat{\mathbf{A}}_{t+1} \right) \left(\sum_{t=1}^{T-1} \mathbf{a}_{t} \mathbf{a}_{t}' - \sum_{t=1}^{T-1} \widehat{Var} \left(\mathbf{u}_{\mathbf{a},t} \right) \right)^{-1},$$

where

$$\mathbf{a}_{t} \equiv \begin{bmatrix} 1_{[z_{t} \ge c]} \\ 1 - 1_{[z_{t} \ge c]} \\ \hat{\mathbf{x}}_{t}^{(1)} \\ \hat{\mathbf{x}}_{t}^{(2)} \end{bmatrix} \qquad \mathbf{u}_{\mathbf{a},t} \equiv \begin{bmatrix} 0 \\ 0 \\ \mathbf{u}_{t}^{(1)} \\ \mathbf{u}_{t}^{(2)} \\ \mathbf{u}_{t}^{(2)} \end{bmatrix}$$

and $\hat{\mathbf{A}}_{t+1} = \begin{bmatrix} \mathbf{0} & \mathbf{0} & \widehat{Cov}\left(\mathbf{u}_{t+1}^{(1)}, \mathbf{u}_{t}^{(1)}\right) & \widehat{Cov}\left(\mathbf{u}_{t+1}^{(2)}, \mathbf{u}_{t}^{(2)}\right) \end{bmatrix}$. The solution to the last moment condition in (18) is also given in closed form and implies the following estimator

$$\widehat{Var}\left(\mathbf{w}_{t+1}\right) = \frac{1}{T - 1 - 2\left(n_x + 1\right)} \sum_{t=1}^{T-1} \widehat{\mathbf{w}}_{t+1} \widehat{\mathbf{w}}_{t+1}' - \frac{1}{T - 1} \sum_{t=1}^{T-1} \widehat{\mathbf{\Omega}}_{t+1},$$
(19)

where

$$\begin{split} \hat{\mathbf{\Omega}}_{t+1} &= \widehat{Var}\left(\mathbf{u}_{t+1}\right) + \hat{\mathbf{h}}_{\mathbf{x}}^{(1)} \widehat{Var}\left(\mathbf{u}_{t}^{(1)}\right) \left(\hat{\mathbf{h}}_{\mathbf{x}}^{(1)}\right)' + \hat{\mathbf{h}}_{\mathbf{x}}^{(2)} \widehat{Var}\left(\mathbf{u}_{t}^{(2)}\right) \left(\hat{\mathbf{h}}_{\mathbf{x}}^{(2)}\right)' \\ &- \widehat{Cov}\left(\mathbf{u}_{t+1}^{(1)}, \mathbf{u}_{t}^{(1)}\right) \left(\hat{\mathbf{h}}_{\mathbf{x}}^{(1)}\right)' - \hat{\mathbf{h}}_{\mathbf{x}}^{(1)} \widehat{Cov}\left(\mathbf{u}_{t}^{(1)}, \mathbf{u}_{t+1}^{(1)}\right) \\ &- \widehat{Cov}\left(\mathbf{u}_{t+1}^{(2)}, \mathbf{u}_{t}^{(2)}\right) \left(\hat{\mathbf{h}}_{\mathbf{x}}^{(2)}\right)' - \hat{\mathbf{h}}_{\mathbf{x}}^{(2)} \widehat{Cov}\left(\mathbf{u}_{t}^{(2)}, \mathbf{u}_{t+1}^{(2)}\right) \end{split}$$

with $\hat{\mathbf{w}}_{t+1} \equiv \hat{\mathbf{x}}_{t+1} - \mathbf{1}_{\{z_t \ge c\}} \hat{\mathbf{h}}_0^{(1)} - (1 - \mathbf{1}_{\{z_t \ge c\}}) \hat{\mathbf{h}}_0^{(2)} - \hat{\mathbf{h}}_{\mathbf{x}}^{(1)} \hat{\mathbf{x}}_t^{(1)} - \hat{\mathbf{h}}_{\mathbf{x}}^{(2)} \hat{\mathbf{x}}_t^{(2)}$. Note that we adopt a standard degree of freedom correction to the first term in (19) because we estimate $2(n_x + 1)$ unknown parameters per equation in the model. The asymptotic distribution of $\theta_2^{\mathbf{x}}$ for $T \longrightarrow \infty$ follows from Hansen (1982) when applied on the moment conditions in (18). The estimated loadings

in the market prices of risk are then given by $\hat{\lambda}_{0}^{(i)} = \hat{\mathbf{h}}_{0}^{(i)}$ and $\hat{\lambda}_{\mathbf{x}}^{(i)} = \hat{\mathbf{h}}_{\mathbf{x}}^{(i)} - (\mathbf{I} - \hat{\mathbf{\Phi}})$ for $i = \{1, 2\}$. Finally, the standard errors for $\hat{\lambda}_{0}^{(i)}$ and $\hat{\lambda}_{\mathbf{x}}^{(i)}$ are identical to those for $\hat{\mathbf{h}}_{0}^{(i)}$ and $\hat{\mathbf{h}}_{\mathbf{x}}^{(i)}$, respectively. That is, we omit uncertainty about $\hat{\mathbf{\Phi}}$, because this estimator uses Tn_{y} observations and therefore tends faster to infinity than $\hat{\boldsymbol{\theta}}_{2}^{\mathbf{x}}$ when also $n_{y} \longrightarrow \infty$, as noted in Andreasen and Christensen (2015).

A.6 Step 2 of the SR Approach: Inducing Stationarity

If the stability condition for \mathbf{x}_t is not satisfied, we then downscale $\mathbf{I} - \mathbf{\Phi} + \boldsymbol{\lambda}_{\mathbf{x}}^{(i)}$ by δ_i for $i = \{1, 2\}$ if the eigenvalues of $\mathbf{I} - \mathbf{\Phi} + \boldsymbol{\lambda}_{\mathbf{x}}^{(i)}$ are greater than or equal to one. The values of δ_1 and δ_2 are determined as in Andreasen and Meldrum (2014), i.e. by

$$(\delta_1, \delta_2) = \arg\min_{\{\delta_{lower} \le \delta_i < 1\}_{i=1}^2} \sum_{i=1}^{n_x} \left(\frac{\sigma_{i,model}^2\left(\delta_1, \delta_2\right) - \sigma_{i,sample}^2}{\sigma_{i,sample}^2} \right)^2$$

We follow Andreasen and Meldrum (2014) and estimate the unconditional variance of the *i*th pricing factor in the sample from $\{\hat{x}_{i,t}\}_{i=1}^{T}$ using

$$\hat{\sigma}_{i,sample}^{2} = \frac{1}{T-1} \sum_{t=1}^{T} \left(\hat{x}_{i,t} - \hat{E} \left[\hat{x}_{i} \right] \right)^{2} - \frac{1}{T} \sum_{t=1}^{T} \widehat{Var} \left(u_{i,t} \right),$$

where $\hat{E}[\hat{x}_{i,t}] = 1/T \sum_{t=1}^{T} \hat{x}_{i,t}$ and $\widehat{Var}(u_{i,t})$ refers to the estimated variance of $\hat{x}_{i,t}$. The value of the unconditional variance of $x_{i,t}$ in the model is computed by simulation, using

$$\mathbf{x}_{t+1} = \mathbf{1}_{[z_t \ge c]} \boldsymbol{\lambda}_0^{(1)} + (1 - \mathbf{1}_{[z_t \ge c]}) \boldsymbol{\lambda}_0^{(2)} \\ + \left(\delta_1 \mathbf{1}_{[z_t \ge c]} \left\{ \mathbf{I} - \boldsymbol{\Phi} + \boldsymbol{\lambda}_{\mathbf{x}}^{(1)} \right\} + \delta_2 \left(1 - \mathbf{1}_{[z_t \ge c]}\right) \left\{ \mathbf{I} - \boldsymbol{\Phi} + \boldsymbol{\lambda}_{\mathbf{x}}^{(2)} \right\} \mathbf{x}_t + \boldsymbol{\Sigma}_{\mathbf{x}} \boldsymbol{\epsilon}_{\mathbf{x},t+1}^{\mathbb{P}}$$

and (12).

References

- Adrian, T., Crump, R. K. and Moench, E. (2013), 'Pricing the term structure with linear regressions', Journal of Financial Economics 110, 110–138.
- Andreasen, M. M. and Christensen, B. J. (2015), 'The SR approach: A new estimation procedure for nonlinear and non-Gaussian dynamic term structure models', *Journal of Econometrics* 184, 420–451.
- Andreasen, M. M. and Meldrum, A. (2014), 'Dynamic term structure models: the best way to enforce the zero lower bound', *Working Paper*.
- Andreasen, M. M., Møller, S. V. and Sander, M. N. (2016), 'Foreign exchange asymmetries across recessions and expansions', *Working*.
- Andrés, J., López-Salido, J. D. and Nelson, E. (2004), 'Tobin's Imperfect Asset Substitution in Optimizing General Equilibrium', Journal of Money, Credit and Banking 36(4), 665–690.
- Ang, A. and Bekaert, G. (2002), 'Regime switches in interest rates', Journal of Business and Economic Statistics 20, 163–182.
- Ang, A., Bekaert, G. and Wei, M. (2008), 'The term structure of real rates and expected inflation', The Journal of Finance 63(2), 797–849.
- Ang, A. and Piazzesi, M. (2003), 'A no-arbitrage vector autoregression of term structure dynamics with macroeconomic and latent variables', *Journal of Monetary Economics* 50, 745–787.
- Bansal, R. and Zhou, H. (2002), 'Term Structure of Interest Rates with Regime Shifts', Journal of Finance LVII(5), 1997–2043.
- Bauer, M. D. and Hamilton, J. D. (2015), 'Robust Bond Risk Premia', Working paper.
- Bauer, M. D. and Rudebusch, G. D. (2016), 'Resolving the Spanning Puzzle in Macro-Finance Term Structure Models', *Federal Reserve Bank of San Francisco Working Paper*.
- Bekaert, G., Hodrick, R. J. and Marshall, D. A. (1997), 'On bibias in tests of the expectations hypothesis of the term structure of interest rates', *Journal of Financial Economics* 44, 309–348.
- Berge, T. J. and Jordà, O. (2011), 'Evaluating the classification of economic activity into recessions and expansions', American Economic Journal: Macroeconomics 3, 246–277.
- Campbell, J. Y. and Shiller, R. J. (1991), 'Yield spread and interest rate movements: A bird's eye view', The Review of Economic Studies 58(3), 495–514.
- Christensen, J. H. E., Diebold, F. X. and Rudebusch, G. D. (2011), 'The affine arbitrage-free class of Nelson-Siegel term structure models', *Journal of Econometrics* **164**, 4–20.
- Christiansen, C., Eriksen, J. and Møller, S. (2014), 'Forecasting recessions: The role of sentiments', *Journal of Banking and Finance* **49**, 459–468.
- Cochrane, J. H. and Piazzesi, M. (2005), 'Bond risk premia', American Economic Review 95(1), 138–160.
- Cochrane, J. H. and Piazzesi, M. (2008), 'Decomposing the yield curve', Working Paper.
- Dai, Q. and Singleton, K. J. (2002), 'Expectation puzzles, time-varying risk premia and affine models of the term structure', *Journal of Financial Economics* 63, 415–441.
- Dai, Q., Singleton, K. J. and Yang, W. (2004), 'Predictability of Bond Risk Premia and Affine Term Structure Models', Working Paper.
- Dai, Q., Singleton, K. J. and Yang, W. (2007), 'Regime Shifts in a Dynamic Term Structure Model of U.S. Treasury Bond Yields', *Review of Financial Studies* 20(5), 1669–1706.
- Diebold, F. X. and Li, C. (2006), 'Forecasting the term structure of government bond yields', *Journal of Econometrics* 130.

- Diebold, F. X., Rudebusch, G. D. and Aruoba, S. B. (2006), 'The macroeconomy and the yield curve: A dynamic latent factor approach', *Journal of Econometrics* 131, 309–338.
- Duffee, G. R. (2002), 'Term premia and interest rate forecasts in affine models', *The Journal of Finance* 57(1), 405–443.
- Duffee, G. R. (2011), 'Information in (and not in) the term structure', *Review of Financial Studies* 24(9), 2895–2934.
- Duffee, G. R. (2013), 'Bond pricing and the macroeconomy', Handbook of Economics and Finance pp. 907–967.
- Fama, E. F. (1976), 'Forward rates as predictors of future spot rates', Journal of Financial Economics 3, 361–377.
- Fama, E. F. and Bliss, R. R. (1987), 'The information in long-maturity forward rates', The American Economic Review 77(4), 680–692.
- Fontaine, J.-S. and Garcia, R. (2012), 'Bond liquidity premia', Review of Financial Studies 25, 1207–1254.
- Gerler, M. and Karadi, P. (2013), 'QE 1 vs. 2. vs. 3. ...: A Framework for Analyzing Large-Scale Asset Purchases as a Monetary Policy Tool', *International Journal of Central Banking* 9(S1), 5–53.
- Gray, S. F. (1996), 'Modeling the conditional distribution of interest rates as a regime-switching process', Journal of Financial Economics 42, 27–62.
- Gürkaynak, R., Sack, B. and Wright, J. (2007), 'The U.S. Treasury yield curve: 1961 to the present', *Journal of Monetary Economics* 54, 2291–2304.
- Hamilton, J. D. (1988), 'Rational-expectations econometric analysis of changes in regime: An investigation of the term structure of interest rates', *Journal of Economic Dynamic and Control* **12**, 385–423.
- Hamilton, J. D. and Wu, J. C. (2012), 'Identification and estimation of Gaussian affine term structure models', *Journal of Econometrics* 168, 315–331.
- Hamilton, J. and Wu, J. C. (2014), 'Testable implications of affine-term-structure models', Journal of Econometrics 178, 231–242.
- Hansen, L. P. (1982), 'Large sample properties of generalized method of moments estimators', *Econometrica* 50(4), 1029–1054.
- Joslin, S., Priebsch, M. A. and Singleton, K. J. (2014), 'Risk Premiums in Dynamic Term Structure Models with Unspanned Macro Risks', *The Journal of Finance* LXIX(3), 1197–1233.
- Joslin, S., Singleton, K. J. and Zhu, H. (2011), 'A new perspective on Gaussian dynamic term structure models', The Review of Financial Studies 24, 926–970.
- Ludvigson, S. and Ng (2009), 'Macro factors in bond risk premia', Review of Financial Studies 22, 5027–5067.
- Lynch, A. W. and Wachter, J. (2013), 'Using samples of unequal length in generalized method of moments estimation', *Journal of Financial and Quantitative Analysis* **48**, 277 307.
- Politis, D. N. and White, H. (2004), 'Automatic block-length selection for the dependent bootstrap', Econometric Reviews 23, 53–70.
- Priebsch, M. A. (2013), 'Computing arbitrage-free yields in multi-factor Gaussian shadow-rate term structure models', Federal Reserve Board Finance and Economics Discussion Series 2013-63.
- Pötscher, B. M. and Prucha, I. (1997), Dynamic Nonlinear Econometric Models: Asymptotic Theory, Springer.
- Woodford, M. (2003), 'Interest and prices', Princeton University Press.

Table 1: Wald tests: Campbell-Shiller Regressions

The table reports asymptotic and bootstrap Wald statistics denoted W_{as} and W_{boot} , respectively. Asymptotic *p*-values are in parentheses and bootstrap *p*-values in brackets. The dataset is unsmoothed Fama-Bliss interest rates. Panel A reports results for the full sample period from 1961:06 to 2013:12. Here we use the Lynch and Wachter (2013) adjusted moments estimator as the panel of bond yields is unbalanced. For the two subperiods in panels B and C, we have observations on all maturities and use standard GMM estimation. For these subperiods, we both report results using a standard asymptotic GMM covariance matrix (Newey-West) and a small sample covariance matrix. The small sample estimates are based on a stationary bootstrap where we resample blocks of the left and right hand side of the Campbell-Shiller regressions with optimal average block size using the Politis and White (2004) approach. When drawing the blocks, we make sure to preserve cross-sectional correlations by resampling from the same time points for all maturities. The covariance matrix is computed using a matrix of 100,000 simulated coefficient estimates. Further details are provided in the Online Appendix.

	PN	ſΙ	NBER		
H_0	W_{as}	W_{boot}	W_{as}	W_{boot}	
Panel A: 1961:06-2013:12					
$\alpha_{REC}^{(k)} = \alpha_{EXP}^{(k)}$	389.6(0.00)		202.7 (0.00)		
$eta_{REC}^{(k)} = eta_{EXP}^{(k)}$	283.2(0.00)		243.8(0.00)		
$\alpha_{REC}^{(k)} = \alpha_{EXP}^{(k)}, \beta_{REC}^{(k)} = \beta_{EXP}^{(k)}$	$1035.1 \ (0.00)$		$650.3 \ (0.00)$		
Panel B: 1973:01-2013:12					
$\alpha_{REC}^{(k)} = \alpha_{EXP}^{(k)}$	344.1(0.00)	$156.2 \ [0.00]$	269.3(0.00)	110.8 [0.00]	
$eta_{REC}^{(k)} = eta_{EXP}^{(k)}$	312.4(0.00)	$117.1 \ [0.00]$	282.3(0.00)	$125.8 \ [0.00]$	
$\alpha_{REC}^{(k)} = \alpha_{EXP}^{(k)}, \beta_{REC}^{(k)} = \beta_{EXP}^{(k)}$	1137.7 (0.00)	386.8 [0.00]	808.8(0.00)	$318.2\ [0.00]$	
Panel C: 1983:01-2013:12					
$\alpha_{REC}^{(k)} = \alpha_{EXP}^{(k)}$	784.2(0.00)	$277.4 \ [0.00]$	492.8(0.00)	$109.8 \ [0.00]$	
$\beta_{REC}^{(k)} = \beta_{EXP}^{(k)}$	468.9(0.00)	$140.2 \ [0.00]$	493.1 (0.00)	$95.5 \ [0.00]$	
$\alpha_{REC}^{(k)} = \alpha_{EXP}^{(k)}, \beta_{REC}^{(k)} = \beta_{EXP}^{(k)}$	3333.3(0.00)	$675.4 \ [0.00]$	2544.9(0.00)	$349.5\ [0.00]$	

Table 2: Return Predictability by the Yield Spread: RobustnessPanel A shows the results of regressing excess returns on a 10-year Treasury bond from Global Financial Data on the 10-year yield spread, whereas Panel B shows the results of regressing excess returns on a 20-year Treasury bond from Ibbotson on the 20-year yield spread obtained from the Goyal and Welch dataset. Excess returns are in both panels computed for the next three months with recessions and expansions identified using the NBER recession dates. Newey-West standard errors with 4 lags are provided in parentheses. T is the total number of observations and T_{BEC} is the number of recession months.

in parentineses. 1 is t	the total number	01 01 00001	vaulons a	IG I REC	10 0110 1101		50001011 III	0110110.	
Sample period	T	N	o switch			Recession	s vs. exp	ansions	
	(T_{REC})	μ	θ	R^2	μ_{EXP}	μ_{REC}	θ_{EXP}	θ_{REC}	R^2
Panel A: 10-year	oond returns								
1926:1-2013:12	1053	-0.006	0.70	5.3%	-0.010	0.012	0.83	0.05	7.7%
	(199)	(0.003)	(0.16)		(0.003)	(0.008)	(0.17)	(0.56)	
1926:1-1961:5	422	-0.002	0.37	4.5%	-0.007	0.014	0.58	-0.24	14.5%
	(116)	(0.002)	(0.14)		(0.002)	(0.004)	(0.14)	(0.29)	
1961:6-2013:12	628	-0.007	0.83	5.8%	-0.010	0.012	0.91	0.29	7.5%
	(83)	(0.004)	(0.22)		(0.004)	(0.016)	(0.22)	(1.01)	
			. ,			. ,	. ,	. ,	
Panel B: 20-year b	oond returns								
1926:1-2013:12	1053	-0.007	0.73	5.0%	-0.014	0.021	0.98	-0.42	9.2%
	(199)	(0.003)	(0.18)		(0.003)	(0.009)	(0.19)	(0.50)	
		. ,	. ,			. ,	. ,	. ,	
1926:1-1961:5	422	-0.001	0.36	2.3%	-0.008	0.019	0.61	-0.36	11.8%
	(116)	(0.003)	(0.19)		(0.003)	(0.005)	(0.19)	(0.37)	
	~ /		. /			. ,	. ,	. /	
1961:6-2013:12	628	-0.009	0.85	5.8%	-0.016	0.024	1.08	-0.45	9.2%
	(83)	(0.005)	(0.23)		(0.004)	(0.016)	(0.22)	(0.76)	

Table 3: Estimation Results: Risk-Neutral Coefficients and Dynamics of PMI Given the low estimates of $\Phi(1, 1)$, the value of α is unidentified and set to 0 for all models. Asymptotic standard errors for $\hat{\Phi}$ in brackets are robust to the measurement errors $v_{t,k}$ displaying heteroskedasticity in the time series dimension, cross-sectional correlation, and autocorrelation. We use $w_D = 5$ and $w_T = 10$ in the provided estimator of Andreasen and Christensen (2015). The asymptotic standard errors for $\hat{\Sigma}_{\mathbf{x}}$ and the parameters for PMI are computed as described in Appendix A.5 and reported in parentheses. Significance at the 10 and 5 percent level is denoted by * and **, respectively.

	\mathcal{M}^{ATSM}	$\mathcal{M}_{\lambda_0}^{ATSM}$	$\mathcal{M}_{\lambda_{r}}^{ATSM}$	$\mathcal{M}^{ATSM}_{\lambda_0\lambda_x}$
$\Phi\left(1,1\right)$	1.18×10^{-8}	1.54×10^{-8}	1.19×10^{-8}	1.28×10^{-8}
$\Phi\left(2,2\right)$	$\begin{array}{c} 0.0352^{**} \\ (0.0046) \end{array}$	0.0353^{**} (0.0046)	$\begin{array}{c} 0.0309^{**} \\ (0.0054) \end{array}$	0.0352^{**} (0.0046)
$\Phi\left(3,3 ight)$	0.0416^{**} (0.0044)	0.0416^{**} (0.0044)	0.0350^{**} (0.0047)	0.0416^{**} (0.0044)
$\Sigma_{\mathbf{x}}(1,1)$	$3.64 \times 10^{-4**}$ (1.97×10 ⁻⁵)	$3.64 \times 10^{-4**}$ (1.96×10 ⁻⁵)	$5.39 \times 10^{-4**}$ (2.46×10 ⁻⁵)	$3.65 \times 10^{-4**}$ (1.94×10 ⁻⁵)
$\Sigma_{\mathbf{x}}(2,1)$	-0.0034^{**} (3.61×10 ⁻⁴)	-0.0034^{**} (3.63×10 ⁻⁴)	-0.0049^{**} (5.32×10 ⁻⁴)	-0.0034^{**} (3.57×10 ⁻⁴)
$\Sigma_{\mathbf{x}}\left(2,2 ight)$	0.0042^{**} (1.69×10 ⁻⁴)	0.0042^{**} (1.71×10 ⁻⁴)	0.0068^{**} (2.51×10 ⁻⁴)	0.0041^{**} (1.68×10 ⁻⁴)
$\Sigma_{\mathbf{x}}\left(3,1\right)$	0.0031^{**} (3.45×10 ⁻⁴)	0.0031^{**} (3.48×10 ⁻⁴)	0.0045^{**} (5.09×10 ⁻⁴)	0.0031^{**} (3.42×10 ⁻⁴)
$\Sigma_{\mathbf{x}}\left(3,2 ight)$	-0.0041^{**} (1.71×10 ⁻⁴)	-0.0041^{**} (1.72×10 ⁻⁴)	-0.0066^{**} (2.49×10 ⁻⁴)	-0.0041^{**} (1.69×10 ⁻⁴)
$\Sigma_{\mathbf{x}}\left(3,3 ight)$	$3.63 \times 10^{-4**} \\ _{(4.05 \times 10^{-5})}$	$3.51 \times 10^{-4**} \\ (3.66 \times 10^{-5})$	$\begin{array}{c} 4.06 \times 10^{-4**} \\ (1.98 \times 10^{-5}) \end{array}$	$3.40 \times 10^{-4**} \\ (2.24 \times 10^{-5})$
PMI				
γ_0	_	0.0094^{**} (0.0043)	0.0095^{**} (0.0044)	0.0094^{**} (0.0043)
γ_z	_	0.9338^{**} (0.0180)	0.9338^{**} (0.0180)	0.9338^{**} (0.0180)
$\gamma_{x}\left(1,1\right)$	_	-1.4022^{**} (0.6955)	-1.3459^{**} (0.6775)	-1.4023^{**} (0.6955)
$\gamma_{x}\left(2,1\right)$	_	-3.1301^{**} $_{(0.6795)}$	-2.9007^{**} (0.6413)	-3.1288^{**} $_{(0.6793)}$
$\gamma_{x}\left(3,1\right)$	—	-3.4533^{**} $_{(0.7357)}$	-3.1354^{**} (0.6791)	-3.4552^{**} (0.7361)
Σ_{zz}	_	0.0222^{**} (8.87×10 ⁻⁴)	0.0222^{**} (8.87×10 ⁻⁴)	0.0222^{**} (8.87×10 ⁻⁴)

recessions in $\mathcal{M}_{\lambda_0}^{ATSM}$.] 10 and 5 percent level is	The asymptotic sta denoted by * and	andard errors are (**, respectively.	computed as desc	cribed in App	endix A.5 and	l reported in par	entheses. Significan	nce at the
	M_{MSLW}	$M_{\lambda 0}^{AT}$	WS	$M^{AT}_{\lambda_{m}}$	MSJ	$M^{A7}_{\lambda 0}$	ΓSM	
		EXP	REC	EXP	REC	EXP	REC	
Intercepts:								
$\lambda_{0}\left(1,1 ight)$	$\begin{array}{c} 2.37 \times 10^{-4**} \\ (5.74 \times 10^{-5}) \end{array}$	$2.29 imes 10^{-4**}$ (5.78×10 ⁻⁵)	$1.64 imes 10^{-4*}$ (9.01×10 ⁻⁵)	0.00 (1.17×	10^{**} $10^{-4})$	$2.74 imes 10^{-4**}$ (6.04×10 ⁻⁵)	$6.20 imes 10^{-4*}$ $_{(3.68 imes 10^{-4})}$	
$\lambda_{0}\left(2,1\right)$	-0.0033^{**} (9.26×10 ⁻⁴)	-0.0032^{**} (9.33×10 ⁻⁴)	-0.0019 (0.0014)	-0.0(359^{**}	-0.0031^{**} (9.73×10 ⁻⁴)	-0.0035 (0.0047)	
$\lambda_{0}\left(3,1 ight)$	0.0032^{**} (9.03×10^{-4})	0.0030^{**} $(9.10{\times}10^{-4})$	0.0015 (0.0014)	0.00	53^{**}_{116}	0.0029^{**} (9.53×10^{-4})	0.0026 (0.0045)	
Slopes:								
$\lambda_{\mathbf{x}} \left(1, 1 ight)$	-0.0316^{**} (0.0094)	-0.02	93^{**}	-0.1651^{**} (0.0210)	-0.0517^{**} (0.0256)	-0.0392^{**} (0.0104)	-0.0778^{*} (0.0438)	
$\lambda_{\mathbf{x}} \left(1, 2 \right)$	$0.0224^{**}_{(0.0087)}$	0.022	11** 83)	0.0217 (0.0179)	0.0244 (0.0690)	$0.0227^{**}_{(0.0094)}$	0.0388 (0.0495)	
$\lambda_{\mathbf{x}} (1,3)$	0.0188^{**} (0.0093)	0.018 (0.000)	55** 89)	$\begin{array}{c} 0.0187 \\ (0.0192) \end{array}$	$\begin{array}{c} 0.0191 \\ (0.0722) \end{array}$	0.0196^{st} (0.0101)	$\begin{array}{c} 0.0305 \\ (0.0478) \end{array}$	
$\lambda_{\mathbf{x}}(2,1)$	0.5096^{**} (0.1624)	0.466	90) (08**	$0.6461^{**}_{(0.2799)}$	0.8623^{**} (0.2401)	0.4954^{**} (0.1812)	$0.3994 \\ (0.5612)$	
$\lambda_{\mathbf{x}}(2,2)$	$\begin{array}{c} 0.0721 \\ (0.1762) \end{array}$	0.07	$72 \\ 54)$	-0.0742 (0.2887)	-0.4898 (0.6264)	$\begin{array}{c} 0.1024 \\ (0.1840) \end{array}$	-0.3539 (0.6061)	
$\lambda_{\mathbf{x}}\left(2,3 ight)$	$0.1546 \\ (0.1903)$	0.16	$\begin{array}{c} 0.2\\ 0.0 \end{array}$	$\begin{array}{c} 0.1282 \\ (0.3013) \end{array}$	-0.3804 (0.6756)	$\begin{array}{c} 0.1958 \\ (0.1979) \end{array}$	-0.2579 (0.6314)	
$\lambda_{\mathbf{x}}\left(3,1 ight)$	-0.4976^{**} (0.1602)	-0.44	55** 70)	-0.6015^{**} (0.2773)	-0.9047^{**} (0.2355)	-0.4644^{**} (0.1804)	-0.3543 (0.5433)	
$\lambda_{\mathbf{x}}$ (3, 2)	-0.0710 (0.1712)	-0.0'	771 05)	-0.0363 (0.2834)	0.2598 (0.6094)	-0.0918 (0.1817)	0.0486 (0.5736)	
$\lambda_{\mathbf{x}}\left(3,3 ight)$	-0.1461 (0.1848)	-0.1	528 47)	-0.2314 (0.2966)	$\begin{array}{c} 0.1416 \\ (0.6559) \end{array}$	-0.1761 (0.1956)	-0.0547 (0.5983)	

Estimates of λ_0 are common across expansions (EXP) and recessions (REC) in $\mathcal{M}_{\lambda_x}^{ATSM}$, while estimates of λ_x are common across expansions and

35

Table 5: Slope Coefficients in Modified Mincer-Zarnowitz Regressions This table reports the regression slopes of modified Mincer-Zarnowitz regressions of realized three months excess returns on a constant and model-implied expected excess returns when conditioning on the state of the economy by using the NBER recession dates. Excess returns are pooled within bins of two years for these regressions, given a total of eight excess returns per reported maturity range. Newey-West standard errors with 6 lags are provided in parentheses

circles with 0 lags	are provi	lucu in p	aremunes	<i>.</i>							
		Е	xpansio	ns		Recessions					
		Maturit	y range	in years	:	Maturity range in years:					
	0-2	2-4	4-6	6-8	8-10	0-2	2-4	4-6	6-8	8-10	
\mathcal{M}^{ATSM}	$\begin{array}{c} 0.976 \\ \scriptscriptstyle (0.115) \end{array}$	1.163 (0.094)	$1.176 \\ (0.098)$	1.120 (0.106)	$0.997 \\ (0.100)$	0.624 (0.775)	$\underset{(0.565)}{0.390}$	$0.097 \\ (0.526)$	-0.093 (0.551)	-0.050 (0.487)	
$\mathcal{M}^{ATSM}_{\lambda_0\lambda_x}$	$\substack{0.648\\(0.090)}$	$\underset{(0.069)}{0.760}$	$\underset{(0.073)}{0.759}$	$\underset{(0.078)}{0.695}$	$\underset{(0.076)}{0.612}$	$\begin{array}{c} 0.754 \\ (0.103) \end{array}$	$\underset{(0.074)}{0.667}$	$\underset{(0.085)}{0.656}$	$\underset{(0.100)}{0.612}$	$\begin{array}{c} 0.544 \\ (0.102) \end{array}$	
$\mathcal{M}^{SRM}_{\lambda_0\lambda_x}$	$\underset{(0.099)}{0.709}$	$\underset{(0.074)}{0.836}$	$\underset{(0.075)}{0.812}$	$\underset{(0.079)}{0.735}$	$\underset{(0.078)}{0.650}$	$\substack{0.799\\(0.109)}$	$\underset{(0.081)}{0.668}$	$\underset{(0.092)}{0.630}$	$\underset{(0.105)}{0.593}$	$\underset{(0.104)}{0.525}$	
$\mathcal{M}^{AFNS}_{\lambda_0(2,1)\lambda_x(2,2)}$	$\substack{0.775\\(0.103)}$	$\underset{(0.081)}{0.860}$	$\underset{(0.083)}{0.812}$	$\underset{(0.084)}{0.682}$	$\underset{(0.081)}{0.578}$	$\substack{0.710\\(0.138)}$	$\begin{array}{c} 0.707 \\ (0.099) \end{array}$	$\underset{(0.108)}{0.737}$	$\underset{(0.132)}{0.731}$	$\underset{(0.130)}{0.732}$	
\mathcal{M}^{Linear}	$\underset{(0.088)}{0.937}$	$\underset{(0.082)}{1.139}$	$\underset{(0.090)}{1.263}$	$\underset{(0.099)}{1.204}$	$\underset{(0.093)}{1.072}$	$\substack{0.306\\(0.404)}$	$\underset{(0.389)}{0.089}$	-0.063 (0.394)	-0.159 (0.428)	-0.159 (0.391)	
$\mathcal{M}^{Spanned}$	$\underset{(0.087)}{0.927}$	$1.145 \\ (0.082)$	$\underset{(0.090)}{1.275}$	$\underset{(0.099)}{1.222}$	$\underset{(0.095)}{1.118}$	$\underset{(0.392)}{0.326}$	$\underset{(0.381)}{0.109}$	-0.036 (0.387)	-0.119 (0.416)	-0.117 (0.380)	

Table 6: Additional Model Estimates: Risk-neutral Coefficients and Dynamics of PMI Given the low estimates of $\Phi(1, 1)$, the value of α is unidentified and set to 0 for all models. Asymptotic standard errors for $\hat{\Phi}$ in brackets are robust to the measurement errors $v_{t,k}$ displaying heteroskedasticity in the time series dimension, cross-sectional correlation, and autocorrelation. We use $w_D = 5$ and $w_T = 10$ in the provided estimator of Andreasen and Christensen (2015). The asymptotic standard errors for $\hat{\Sigma}_x$ and the parameters for PMI are computed as described in Appendix A.5 and reported in parentheses. For $\mathcal{M}^{Spanned}$, we let $\mu(4, 1) = \Phi(4, i) = \Phi(i, 4) = 0$ for $i = \{1, 2, 3\}$ as the fit of bond yields is completely unaffected by these coefficients. Significance at the 10 and 5 percent level is denoted by * and **, respectively.

	$\mathcal{M}^{SRM}_{\lambda_0\lambda_x}$	$\mathcal{M}^{AFNS}_{\lambda_0(2,1)\lambda_x(2,2)}$	\mathcal{M}^{Linear}	$\mathcal{M}^{Spanned}$
$\beta(4,1)$	_	_	_	0.1283^{**} (0.0204)
${ ilde\lambda}$	_	0.0388^{**} (0.0021)	_	_
$\Phi\left(1,1\right)$	1.00×10^{-8} (3.64×10 ⁻⁴)	_	1.09×10^{-8} (4.44×10 ⁻⁴)	1.11×10^{-8} (6.92×10 ⁻⁴)
$\Phi\left(2,2\right)$	0.0351^{**} (0.0061)	-	0.0352^{**} (0.0046)	$\begin{array}{c} 0.0430 \\ (0.1243) \end{array}$
$\Phi\left(3,3 ight)$	0.0536^{**} $_{(0.0053)}$	_	0.0416^{**} (0.0044)	$\begin{array}{c} 0.0447 \\ (0.1281) \end{array}$
$\Phi\left(4,4\right)$	_	—	—	0.0513^{**} (0.0171)
$\Sigma_{\mathbf{x}}\left(1,1\right)$	$3.44 \times 10^{-4**}$ (1.67×10 ⁻⁵)	$3.67 \times 10^{-4**}$ (1.98×10 ⁻⁵)	$3.65 \times 10^{-4**}$ (3.06×10 ⁻⁵)	$3.16 \times 10^{-4**}$ (2.31×10 ⁻⁵)
$\Sigma_{\mathbf{x}}(2,1)$	-0.0012^{**} (1.36×10 ⁻⁴)	-0.0003^{**} (2.66×10 ⁻⁵)	-0.0034^{**} (5.38×10 ⁻⁴)	-0.0091^{**}
$\Sigma_{\mathbf{x}}\left(2,2\right)$	0.0019^{**} (7.17×10 ⁻⁵)	$3.56 \times 10^{-4**}$ (2.60×10 ⁻⁵)	0.0042^{**} (2.01×10 ⁻⁴)	0.0194^{**} (0.0010)
$\Sigma_{\mathbf{x}}\left(3,1\right)$	$9.68 \times 10^{-4**}$ (1.25×10 ⁻⁴)	-0.0006^{**} (6.06×10 ⁻⁵)	0.0031^{**} (5.12×10 ⁻⁴)	0.0088^{**} (0.0019)
$\Sigma_{\mathbf{x}}\left(3,2 ight)$	-0.0018^{**} (7.55×10 ⁻⁵)	$7.92 \times 10^{-5*}$	-0.0041^{**}	-0.0210^{**}
$\Sigma_{\mathbf{x}}\left(3,3 ight)$	$3.69 \times 10^{-4**}$	$7.05 \times 10^{-4**}$	$3.53 \times 10^{-4**}$	0.0023^{**} (1.07×10^{-4})
$\Sigma_{\mathbf{x}}(4,1)$		(2.31×10)	(0.13×10^{-1}) 6.62×10^{-4} (0.02×10^{-4})	2.71×10^{-4}
$\Sigma_{\mathbf{x}}\left(4,2\right)$	_	_	0.0040^{**}	0.0134^{**}
$\Sigma_{\mathbf{x}}\left(4,3 ight)$	_	_	0.0051^{**}	-0.0176^{**}
$\Sigma_{\mathbf{x}}\left(4,4\right)$	_	_	0.0213^{**}	0.0027^{**} (4.51×10 ⁻⁴)
			(0.01/10)	(4.01×10)
PMI	0.0101**	0.0004**		
۲Ŷ0	(0.0101) (0.0042)	(0.0094)	—	—
γ_z	(0.9320^{**})	0.9338^{**} (0.0180)	—	—
$\gamma_{x}\left(1,1\right)$	-1.4882^{**} (0.6842)	-1.4000^{**} (0.6948)	_	_
$\gamma_{x}\left(2,1\right)$	-3.1238^{**} (0.6649)	-3.2161^{**} (0.6911)	—	—
$\gamma_{x}\left(3,1\right)$	-3.8735^{**}	1.8307^{**}	_	-
Σ_{zz}	$\begin{array}{c} 0.0222^{**} \\ (8.88 \times 10^{-4}) \end{array}$	0.0221^{**} (8.90×10 ⁻⁴)	-	-

	$\mathcal{M}_{\lambda}^{S}$	$\frac{RM}{n\lambda_{r}}$	$\mathcal{M}^{AFN}_{\lambda_0(2)}$	$\frac{1}{NS}$	\mathcal{M}^{Li}	inear	\mathcal{M}^{Sp}	anned
	EXP	REC	EXP	REC	EXP	REC	EXP	REC
Intercepts:								
$\lambda_{0}\left(1,1 ight)$	$2.33 \times 10^{-4**}$ (5.23×10 ⁻⁵)	$5.84 \times 10^{-4*}$ (3.40×10 ⁻⁴)	$3.17 \times (5.84)$	10^{-4**} ×10 ⁻⁵)	$2.36 \times (7.24 \times$	10^{-4**} 10^{-5})	$1.96 \times (6.43 \times$	10^{-4**} 10^{-5})
$\lambda_{0}\left(2,1\right)$	-0.0012^{**} (3.50×10 ⁻⁴)	-0.0014 (0.0017)	-0.0002^{**} (7.56×10 ⁻⁵)	-0.0007^{**} (2.67×10 ⁻⁴)	-0.00)30**)10)	-0.0)064 040)
$\lambda_{0}\left(3,1 ight)$	0.0011^{**} (3.41×10 ⁻⁴)	6.30×10^{-4}	-0.0 (1.56	0005^{**} ×10 ⁻⁴)	0.00 (9.75×	$27^{**}_{10^{-4}}$	0.00 (0.00)52)44)
$\lambda_{0}\left(4,1 ight)$	_	_	_	,	0.00	94** 944)	0.00	070 047)
Slopes:								
$\lambda_{\mathbf{x}}\left(1,1 ight)$	-0.0338^{**} (0.0093)	$-0.0728^{*}_{(0.0408)}$	-0.0 (0.0)468**)097)	-0.03 (0.0)	315** 113)	-0.02 (0.00)	237** 099)
$\lambda_{\mathbf{x}}\left(1,2\right)$	0.0206^{**} (0.0085)	$\begin{array}{c} 0.0376 \\ (0.0457) \end{array}$	0.02	200** 0093)	0.02	24** 098)	0.02	30** 085)
$\lambda_{\mathbf{x}}\left(1,3 ight)$	$\substack{0.0156\\(0.0103)}$	$\underset{(0.0414)}{0.0183}$	0.02	214** 0101)	0.01 (0.01	.88* 105)	0.02	24** 087)
$\lambda_{\mathbf{x}}\left(1,4\right)$	_	_	_	_	$6.44 \times (1.75 \times$	$(10^{-6})^{-4}$	0.00	26** 012)
$\lambda_{\mathbf{x}}\left(2,1\right)$	0.2008^{**} (0.0666)	$\underset{(0.2103)}{0.1186}$	0.03	326^{**}	0.48	88^{**} 179)	$1.13_{(0.55)}$	$12^{**}_{562)}$
$\lambda_{\mathbf{x}}\left(2,2\right)$	$0.0889 \\ (0.0723)$	-0.2956 $_{(0.2527)}$	$\underset{(0.0138)}{0.0081}$	-0.2363^{**} $_{(0.1183)}$	0.0' (0.18	734 346)	-0.8 (0.73	3757 308)
$\lambda_{\mathbf{x}}\left(2,3 ight)$	0.1766^{**} $_{(0.0854)}$	-0.2380 (0.2903)	-0.0 (0.0)404**)156)	0.13 (0.19	550 960)	-0.8 (0.74	$3365_{401)}$
$\lambda_{\mathbf{x}}\left(2,4 ight)$	_	_	_	_	-0.0 (0.00)024)31)	-0.1 (0.10	1060 024)
$\lambda_{\mathbf{x}}\left(3,1 ight)$	-0.1801^{**} (0.0674)	-0.0880 (0.1972)	0.08 (0.0	853** 0279)	-0.46	658** 152)	-0.9 (0.60)329 010)
$\lambda_{\mathbf{x}}\left(3,2 ight)$	-0.0821 (0.0722)	-0.0039 (0.2155)	0.0 (0.0)179)305)	-0.0)731 ⁷⁸¹⁾	1.32	$262^{*}_{577)}$
$\lambda_{\mathbf{x}}\left(3,3 ight)$	-0.1516^{*} (0.0853)	-0.0769 (0.2431)	-0.0)925**)264)	-0.1 (0.18)	469 394)	1.30 (0.7)	000* 773)
$\lambda_{\mathbf{x}}\left(3,4\right)$	_	_	_	_	0.00)37)30)	0.1' (0.10	738 074)
$\lambda_{\mathbf{x}}\left(4,1\right)$	_	_	_	_	-1.40 (0.69))20** 978)	-1.44 (0.7)	433** 166)
$\lambda_{\mathbf{x}}\left(4,2 ight)$	_	—	_	_	-3.12 (0.66	287** 590)	-3.48 (0.72)	870** 216)
$\lambda_{\mathbf{x}}\left(4,3 ight)$	_	—	_	_	-3.45	530** 086)	-3.57	736** 326)
$\lambda_{\mathbf{x}}\left(4,4\right)$	_	-	_	_	-0.00	662 ^{**} 164)	-0.50 (0.09)74 ^{**} 981)

Table 7: Additional Model Estimates: The Market Prices of Risk The asymptotic standard errors are computed as described in Appendix A.5 and reported in parentheses. Significance at the 10 and 5 percent level is denoted by * and **, respectively.

Figure 1: Ordinary and Modified Campbell-Shiller Regressions

Estimation results from ordinary and modified Campbell-Shiller regressions are provided in the upper and lower part of the figure, respectively, using monthly data from 1961:6 to 2013:12. The dataset is unsmoothed Fama-Bliss (FB) interest rates or the data companying Gürkaynak, Sack and Wright (2007), denoted GSW. Intercepts are multiplied by 100.



Figure 2: Goodness-of-fit in Ordinary and Modified Campbell-Shiller Regressions The goodness of fit as measured by the R^2 statistic (in percent) for the ordinary and modified Campbell-Shiller regressions using monthly data from 1973:1 to 2013:12. The dataset is unsmoothed Fama-Bliss (FB) interest rates or the data companying Gürkaynak, Sack and Wright (2007), denoted GSW.



Figure 3: Ordinary and Modified Return Regressions

These charts report the slope coefficients in return regressions where excess bond returns three months ahead are regressed on either the yield spread, the forward spread, or the CP factor (with a constant) using monthly data from 1961:6 to 2013:12. The dataset is unsmoothed Fama-Bliss (FB) interest rates or the data companying Gürkaynak, Sack and Wright (2007), denoted GSW. Given the well-known reliance of the

CP factor on FB forward rates, the CP factor for the GSW dataset is computed as $cp_t^{GSW} = (\hat{\gamma}^{GSW})' F_t$, where $F_t \equiv \begin{bmatrix} 1, y_{t,12}, f_t^{(12,24)}, f_t^{(24,36)}, f_t^{(36,48)}, f_t^{(48,60)} \end{bmatrix}'$ are FB forward rates but $\hat{\gamma}^{GSW}$ is obtained by regressing GSW returns on FB forward rates.



Figure 4: Model Evaluation: Ordinary and Modified Campbell-Shiller Regressions The empirical intercepts and slope coefficients in ordinary and modified Campbell-Shiller regressions are computed based on monthly unsmoothed Fama-Bliss interest rates from 1961:6 to 2013:12. The model-implied regression loadings are computed at the estimated parameters for $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ using a simulated sample path of 100,000 observations. Intercepts are multiplied by 100.



Figure 5: Model Evaluation: Ordinary and Modified return Regressions The empirical slope coefficients in ordinary and modified return regressions are computed based on monthly unsmoothed Fama-Bliss interest rates from 1961:6 to 2013:12. The model-implied regression loadings are computed at the estimated parameters for $\mathcal{M}_{\lambda_0\lambda_x}^{ATSM}$ using a simulated sample path of 100,000 observations.



Figure 6: Excess Returns and Term Premia: The Gaussian ATSM vs. Regime-Dependent ATSM The top chart reports quarterly expected excess returns for the 10-year bond when expressed in annualized percent. Term premia (in percent) at maturity k is defined as $y_{t,k} - \frac{1}{k} \sum_{i=0}^{k-1} \mathbb{E}_t \left[r_{t+i} \right]$. Both measures of bond risk premia are computed at the estimated states, i.e. $\hat{\mathbf{x}}_t \left(\hat{\boldsymbol{\theta}}_{11}^{step3}, \hat{\boldsymbol{\Sigma}}^{step3} \right)$ With regime switching, expected excess returns and term premia are computed by Monte Carlo integration using 10,000 draws. Shaded areas denote NBER recessions.



Figure 7: SRM with Regime-Switching: Ordinary and Modified Campbell-Shiller Regressions The empirical intercepts and slope coefficients in ordinary and modified Campbell-Shiller regressions are

The empirical intercepts and slope coefficients in ordinary and modified Campbell-Shiller regressions are computed based on monthly unsmoothed Fama-Bliss interest rates from 1961:6 to 2013:12. The model-implied regression loadings are computed at the estimated parameters for $\mathcal{M}_{\lambda_0\lambda_x}^{SRM}$ using a simulated sample path of 100,000 observations. Intercepts are multiplied by 100.



Figure 8: AFNS Model with Regime-Switching: Ordinary and Modified Campbell-Shiller Regressions

The empirical intercepts and slope coefficients in ordinary and modified Campbell-Shiller regressions are computed based on monthly unsmoothed Fama-Bliss interest rates from 1961:6 to 2013:12. The model-implied regression loadings are computed at the estimated parameters for $\mathcal{M}_{\lambda_0(2,1)\lambda_x(2,2)}^{AFNS}$ using a simulated sample path of 100,000 observations. Intercepts are multiplied by 100.







Figure 10: Linear and Spanned Models: Ordinary and Modified Campbell-Shiller Regressions

The empirical intercepts and slope coefficients in ordinary and modified Campbell-Shiller regressions are computed based on monthly unsmoothed Fama-Bliss interest rates from 1961:6 to 2013:12. The model-implied regression loadings are computed at the estimated parameters for \mathcal{M}_{Linear} and $\mathcal{M}_{Spanned}$ using a simulated sample path of 100,000 observations. Intercepts are multiplied by 100.



Research Papers 2016



- 2016-09: Ole E. Barndorff-Nielsen: Assessing Gamma kernels and BSS/LSS processes
- 2016-10: Tim Bollerslev, Andrew J. Patton and Rogier Quaedvlieg: Modeling and Forecasting (Un)Reliable Realized Covariances for More Reliable Financial Decisions
- 2016-11: Tom Engsted and Thomas Q. Pedersen: The predictive power of dividend yields for future inflation: Money illusion or rational causes?
- 2016-12: Federico A. Bugni, Mehmet Caner, Anders Bredahl Kock and Soumendra Lahiri: Inference in partially identified models with many moment inequalities using Lasso
- 2016-13: Mikko S. Pakkanen and Jani Lukkarinen: Arbitrage without borrowing or short selling?
- 2016-14: Andrew J.G. Cairns, Malene Kallestrup-Lamb, Carsten P.T. Rosenskjold, David Blake and Kevin Dowd: Modelling Socio-Economic Differences in the Mortality of Danish Males Using a New Affluence Index
- 2016-15: Mikkel Bennedsen, Ulrich Hounyo, Asger Lunde and Mikko S. Pakkanen: The Local Fractional Bootstrap
- 2016-16: Martin M. Andreasen and Kasper Jørgensen: Explaining Asset Prices with Low Risk Aversion and Low Intertemporal Substitution
- 2016-17: Robinson Kruse, Christian Leschinski and Michael Will: Comparing Predictive Accuracy under Long Memory With an Application to Volatility Forecasting
- 2016-18: Søren Johansen and Bent Nielsen: Tightness of M-estimators for multiple linear regression in time series
- 2016-19: Tim Bollerslev, Jia Li and Yuan Xue: Volume, Volatility and Public News Announcements
- 2016-20: Andrea Barletta, Paolo Santucci de Magistris and Francesco Violante: Retrieving Risk-Neutral Densities Embedded in VIX Options: a Non-Structural Approach
- 2016-21: Mikkel Bennedsen: The Local Fractional Bootstrap
- 2016-22: Søren Johansen and Morten Ørregaard Nielsen: The cointegrated vector autoregressive model with general deterministic terms
- 2016-23: Yunus Emre Ergemen and Carlos Vladimir Rodríguez-Caballero: A Dynamic Multi-Level Factor Model with Long-Range Dependence
- 2016-24: Shin Kanaya: Convergence rates of sums of α-mixing triangular arrays: with an application to non-parametric drift function
- 2016-25: Gustavo Fruet Dias, Marcelo Fernandes and Cristina M. Scherrer: Improving on daily measures of price discovery
- 2016-26: Martin M. Andreasen, Tom Engsted, Stig V. Møller and Magnus Sander: Bond Market Asymmetries across Recessions and Expansions: New Evidence on Risk Premia